

# Analytic Torsion and Dynamical Flow: A Survey on the Fried Conjecture

### Shu Shen

**Abstract.** Given an acyclic and unitarily flat vector bundle on a closed manifold, Fried conjectured an equality between the analytic torsion and the value at zero of the Ruelle zeta function associated to a dynamical flow. In this survey, we review the Fried conjecture for different flows, including the suspension flow, the Morse–Smale flow, the geodesic flow, and the Anosov flow.

Mathematics Subject Classification (2010). 58J20, 58J52, 11F72, 11M36, 37C30. Keywords. Index theory and related fixed point theorems, analytic torsion, Selberg trace formula, Ruelle dynamical zeta function.

### Contents

| Int | roduct | ion  | 248 |
|-----|--------|--|-----|
| 1.  | The    | Ray-Singer analytic torsion                                | 251 |
|     | 1.1.   | The flat vector bundle                                     | 252 |
|     | 1.2.   | Hodge Laplacian and analytic torsion                       | 252 |
|     | 1.3.   | A formula for fibrations                                   | 254 |
| 2.  | The :  | Fried conjecture   | 255 |
|     | 2.1.   | Lefschetz index  | 255 |
|     | 2.2.   | Fuller index   | 257 |
|     | 2.3.   | The Ruelle dynamical zeta function                         | 258 |
| 3.  | Susp   | ension flow of a diffeomorphism                            | 260 |
|     | 3.1.   | The definition of the suspension flow                      | 260 |
|     | 3.2.   | The flat vector bundle on the suspension                   | 261 |
|     | 3.3.   | The Ruelle dynamical zeta function of the suspension flow  | 262 |
| 4.  | Mors   | e–Smale flow   | 264 |
|     | 4.1.   | The definition of Morse–Smale flow                         | 264 |
|     | 4 2    | The Ruelle dynamical zeta function of the Morse-Smale flow | 265 |

The author is indebted to Xiaolong Han, Jianqing Yu, and referees for reading this paper very carefully.

| 5.  | Geodesic flow |   |     |
|-----|---------------|---|-----|
|     | 5.1.          | Flat vector bundle on the unit tangent bundle                         | 268 |
|     | 5.2.          | Geodesic flow of the non-positively curved manifolds                  | 270 |
|     | 5.3.          | The $V$ -invariant and a formal proof of the Fried conjecture $\dots$ | 271 |
|     | 5.4.          | Reductive group, globally and locally symmetric space                 | 272 |
|     | 5.5.          | The Selberg trace formula   | 276 |
|     | 5.6.          | Moscovici–Stanton's vanishing theorem                                 | 280 |
|     | 5.7.          | The case $G = SO^0(p, 1)$ with $p$ odd                                | 281 |
|     | 5.8.          | The reductive group $G$ with $\delta(G) = 1$                          | 284 |
| 6.  |               | ov flow   |     |
|     | 6.1.          | Closed orbits of the Anosov flow                                      | 287 |
|     | 6.2.          | The meromorphic extension   | 288 |
|     | 6.3.          | The $R_{\phi,\rho}(0)$ as a topological invariant                     | 291 |
|     |               | Anosov flow on 3-manifold   |     |
| Ref | ference       | es  | 294 |

### Introduction

The purpose of this survey is to review some recent progress of the Fried conjecture, which affirms an equality between the analytic torsion and the value at zero of the Ruelle dynamical zeta function.

In the first two sections of the paper, we describe the Ray–Singer analytic torsion of a flat vector bundle and the Ruelle zeta function of a dynamical flow. The next three sections are devoted to the study on the Fried conjecture for certain flows, including the suspension flow, the Morse–Smale flow, the geodesic flow, and the Anosov flow.

The paper is written in an informal way. We only sketch the proofs, and refer to the original papers when necessary.

We now describe in more details the content of this paper, and give the proper historical perspective to the results described in the paper.

### 0.1. The combinatorial and analytic torsions

Let Z be a smooth closed manifold. Let F be a complex unitarily flat vector bundle on Z, which amounts to specifying a unitary finite-dimensional representation  $\rho$ of the fundamental group  $\pi_1(Z)$ . Denote by  $H^{\cdot}(Z,F)$  the cohomology of the sheaf of locally constant sections of F. Assume that F is acyclic, i.e.,  $H^{\cdot}(Z,F) = 0$ .

The Reidemeister combinatorial torsion [Re35, Fr35, dR50] is a positive real number defined with the help of a triangulation on Z. However, it does not depend on the triangulation and becomes a topological invariant. It is the first invariant that can distinguish closed manifolds such as lens spaces which are homotopy equivalent but not homeomorphic.

The analytic torsion was introduced by Ray and Singer [RS71] as an analytic counterpart of the Reidemeister torsion. In order to define the analytic torsion one has to choose a Riemannian metric on Z and a Hermitian metric on F. The analytic torsion is a certain weighted alternating product of regularized determinants of the Hodge Laplacians acting on the space of differential forms with values in F.

The celebrated Cheeger–Müller Theorem [C79, M78] tells us that the Ray–Singer analytic torsion coincides with the Reidemeister combinatorial torsion. Bismut–Zhang [BZ92] and Müller [M93] simultaneously considered generalizations of this result. Müller [M93] extended this result to the case where F is unimodular, i.e.,  $|\det \rho(\gamma)| = 1$  for all  $\gamma \in \pi_1(Z)$ . Bismut and Zhang [BZ92, Theorem 0.2] generalised the original Cheeger–Müller Theorem to arbitrary flat vector bundles with arbitrary Hermitian metrics. There are also various extensions to the equivariant case by Lott–Rothenberg [LoRo91], Lück [Lü93], and Bismut–Zhang [BZ94], to the family case by Bismut–Goette [BG01] under the assumption of the existence of the fibrewise Morse function, and to manifolds with boundaries by Brüning–Ma [BrMa13].

### 0.2. Dynamical zeta function

The grandmother of all zeta functions is the Riemann zeta function defined for Re(s) > 1 by

$$\zeta(s) = \sum_{n=1}^{\infty} \frac{1}{n^s}.$$
(0.1)

Riemann showed that  $\zeta(s)$  extends meromorphically to **C** with a single pole at s=1 and that there is a functional equation relating  $\zeta(s)$  and  $\zeta(1-s)$ . The Euler product formula asserts that for Re(s) > 1,

$$\zeta(s) = \prod_{p:\text{prime}} (1 - p^{-s})^{-1}.$$
 (0.2)

It tells us that  $\zeta(s)$  encodes the distribution of the prime numbers. Some statistical properties, like the prime number theorem, can be deduced from the information on the poles and zeros of  $\zeta(s)$ .

After Riemann's work, several zeta functions with similar properties have been introduced. In particular, Weil [W49] constructed a zeta function using the Frobenius map T defined on an algebraic variety Z over a finite field. It counts the closed orbits of the discrete dynamical system (Z,T). To a smooth closed manifold with a diffeomorphism, similar zeta function was also introduced by Artin and Mazur [ArMaz65].

To a flat vector bundle on the underlying manifold, we can associate a weight to the dynamical systems. In [Ru76a], Ruelle introduced his zeta function for a weighted dynamical system of a diffeomorphism or a flow.

For the geodesic flow on the unit tangent bundle of a Riemann surface of genus  $\geq 2$ , an application of the Selberg trace formula [Sel56] shows that the Ruelle dynamical zeta function has a meromorphic extension to  $\mathbf{C}$ . Similar methods can

be generalized to hyperbolic manifolds by Fried [F86a], to the locally symmetric space of rank 1 by Bunke and Olbrich [BuO95], to the locally symmetric space of higher ranks by Moscovich–Stanton [MoSt91], Shen [Sh18], and Shen–Yu [ShY17].

Independent of the above Selberg theory, which is based on the spectral theory of the Laplacian, a thermodynamical formalism, which is based on the spectral theory of the transfer operator, is used to study the Anosov flow (or more generally Axiom A flow¹) by Ruelle [Ru76b]. He showed that if the flow itself and the stable and unstable foliations are all analytic, then his zeta function has a meromorphic extension to  $\bf C$ . Fried [F86c] generalized Ruelle's result by requiring only the flow and the stable foliation to be analytic. An important extension of the above results was given by Rugh [Rug96], for three-dimensional manifolds, and then by Fried [F95], in arbitrary dimensions, but still assuming the analyticity of the flow. The extension of such results to the  $C^{\infty}$  setting was only given very recently by Giulietti, Liverani, and Pollicott [GiLiPo13] and by Dyatlov and Zworski [DyZ16] (See also [FaT17, DyGu16, DyGu18, BWSh20] for related works). This recent progress on the dynamical zeta function is based on the introduction of the anisotropic space. We refer the reader to the book of Baladi [Ba18] for an introduction of these techniques.

### 0.3. The Fried conjecture

It was Milnor [Mi68a] who observed a remarkable similarity between the Reidemeiter torsion and the Weil zeta function. A precise and quantitative description of their relation was obtained by Fried [F86a, F86b] for the geodesic flow on the unit tangent bundle of a hyperbolic manifold. He showed that the analytic/combinatorial torsion of an acyclic unitarily flat vector bundle on a unit tangent bundle of a closed oriented hyperbolic manifold is equal to the value at zero of the Ruelle dynamical zeta function<sup>2</sup> associated to the geodesic flow. He conjectured later [F87, F95] that similar results hold true for more general flows.

Four kinds of flows will be examined in this survey. As a warm up, we will begin with two simple flows: the suspension flow and Morse–Smale flow. As we will see, the Fried conjecture for the suspension flow is just the Lefschetz fixed point formula, and the Fried conjecture for the Morse–Smale flow is a consequence of the Cheeger–Müller/Bismut–Zhang theorem (see [F87, Theorem 3.1] and [ShY18, Theorem 0.2]). Next, we will consider the geodesic flow on the unit tangent bundle of the locally symmetric spaces. In this case, the Fried conjecture can be deduced formally via the V-invariant of Bismut–Goette [BG04]. Following previous contributions by Moscovici–Stanton [MoSt91], a rigorous proof is given by the author [Sh18] using Bismut's orbital integral formula [B11]. In [ShY17], Shen and Yu made a further generalization to closed locally symmetric orbifolds. Finally, we will study the Anosov flow. If the underlying manifold has dimension 3, it was known by Sánchez-Morgado [SM93, SM96a] that the Fried conjecture holds true

<sup>&</sup>lt;sup>1</sup>In this case, the meromorphic extension problem is called the Smale conjecture [Sm67].

<sup>&</sup>lt;sup>2</sup>Here the Ruelle dynamical zeta function should be twisted by the holonomy of the flat vector bundle.

if the flow is transitive and analytic, and if the flat vector bundle satisfies certain holonomy conditions. Using a variation formula, Dang, Guillarmou, Rivière, and Shen [DaGuRiSh20] removed the analyticity assumption in Sánchez-Morgado's result. For general Anosov flow, e.g., the geodesic flow on the unit tangent bundle of a negatively curved manifold, the Fried conjecture is still open. The only known result is that under certain spectral condition on the transfer operator, in [DaGuRiSh20], the authors have shown that the value at zero of the Ruelle dynamical zeta function does not depend on a small perturbation of the Anosov flow.

Due to the limitation of the length of this survey, there are several interesting topics which have not been included. For example, the Fried conjecture for hyperbolic manifolds with cusps [Pa09] or for non-unitarily flat vector bundles [Wo08, M12, M20, Sh20a, Sh20b, Sp20], the relation between the dynamical zeta function with the eta invariant [Mil78, MoSt89, B19] and with the holomorphic torsion [F88, MoSt18]. We refer the reader to the cited references for more information.

### 0.4. Organization of the paper

This paper is organized as follows. In Section 1, we give the main properties of the Ray–Singer analytic torsion.

In Section 2, we introduce the dynamical zeta function and state the Fried conjecture.

In Sections 3–6, we discuss the Fried conjecture for the suspension flow, the Morse–Smale flow, the geodesic flow, and the Anosov flow.

Throughout the paper, we use the supersymmetric convention. For a matrix A acting on a **Z**-graded space E, the supertrace of A is defined by

$$\operatorname{Tr}_{\mathbf{s}}^{E^{\cdot}}[A] = \operatorname{Tr}^{E^{\cdot}}[(-1)^{N}A],$$
 (0.3)

where N is the number operator on E which sends  $e \in E^q$  to  $qe \in E^q$ . We use the notation

$$\mathbf{R}_{+} = [0, \infty), \quad \mathbf{R}_{+}^{*} = (0, \infty), \quad \mathbf{N} = \{0, 1, 2, \ldots\}, \quad \mathbf{N}^{*} = \{1, 2, 3, \ldots\}.$$
 (0.4)

For a finite set A, we denote by |A| the cardinality of A. By a closed orbit we mean a non-trivial closed orbit, i.e., its period is positive.

## 1. The Ray-Singer analytic torsion

The purpose of this section is to recall the definition of the Ray–Singer analytic torsion [RS71] of a Hermitian flat vector bundle on a closed Riemannian manifold.

This section is organized as follows. In Section 1.1, we introduce the flat vector bundle.

In Section 1.2, we recall the definition of the Ray–Singer analytic torsion.

In Section 1.3, we consider a fibration  $M \to Z$  of closed Riemannian manifolds. We give a formula relating the analytic torsions of a Hermitian flat vector bundle on M and of its direct image on Z, which is equipped with an  $L^2$  Hermitian metric.

#### 1.1. The flat vector bundle

Let Z be a smooth closed manifold. Let F be a complex flat vector bundle on Z with flat connection  $\nabla^F$ . Equivalently, F can be obtained via a finite-dimensional complex representation<sup>3</sup>  $\rho: \pi_1(Z) \to \mathrm{GL}_r(\mathbf{C})$  of the fundamental group  $\pi_1(Z)$  of Z. The flat vector bundle F is called unitarily flat if  $\rho$  is unitary.

Let  $(\Omega^{\cdot}(Z, F), d^{Z})$  be the de Rham complex of smooth sections of  $\Lambda^{\cdot}(T^{*}Z) \otimes_{\mathbf{R}} F$  on Z. Let  $H^{\cdot}(Z, F)$  be the cohomology of the above complex. We say that F is acyclic if  $H^{\cdot}(Z, F) = 0$ . The Euler characteristic number of F is then given by

$$\chi(Z,F) = \sum_{q=0}^{\dim Z} (-1)^q \dim H^q(Z,F).$$
 (1.1)

When F is the real trivial line bundle **R**, we use the notation  $(\Omega^{\cdot}(Z), d^{Z}), H^{\cdot}(Z)$  and  $\chi(Z)$ .

**Example 1.1.** Take  $Z = \mathbb{S}^1 = \mathbf{R}/\mathbf{Z}$ . Set  $\alpha \in \mathbf{C}$ . Let F be the trivial complex line bundle  $\mathbf{C}$  equipped with the connection  $\nabla^F = d + \alpha dt$ . The holonomy  $\rho$  is then given by  $\rho : n \in \mathbf{Z} \to e^{n\alpha} \in \mathbf{C}^*$ . It is easy to see that  $H^\cdot(Z, F) = 0$  if and only if  $\alpha \notin 2i\pi\mathbf{Z}$ , and that F is unitarily flat if and only if  $\alpha \in i\mathbf{R}$ .

**Example 1.2.** Let  $0 \to F_0 \to F_1 \to F_2 \to 0$  be an exact sequence of flat vector bundles on Z. Using the associated long exact sequence

$$\cdots \to H^p(Z, F_0) \to H^p(Z, F_1) \to H^p(Z, F_2) \to \cdots, \tag{1.2}$$

we see that if two of  $F_i$  are acyclic, then the third one is also acyclic.

### 1.2. Hodge Laplacian and analytic torsion

Let  $g^{TZ}$  be a Riemannian metric on TZ, and let  $g^F$  be a Hermitian metric on F. For  $s_1, s_2 \in \Omega^{\cdot}(Z, F)$ , put

$$\langle s_1, s_2 \rangle_{\Omega^{\cdot}(Z,F)} = \int_{z \in Z} \langle s_1(z), s_2(z) \rangle_{\Lambda^{\cdot}(T^*Z) \otimes_{\mathbf{R}} F} dv_Z, \tag{1.3}$$

where  $\langle \cdot, \cdot \rangle_{\Lambda^{\cdot}(T^{*}Z) \otimes_{\mathbf{R}} F}$  is the metric on  $\Lambda^{\cdot}(T^{*}Z) \otimes_{\mathbf{R}} F$  induced by  $g^{TZ}, g^{F}$ , and  $dv_{Z}$  is the Riemannian volume of  $(Z, g^{TZ})$ . Then,  $\langle \cdot, \cdot \rangle_{\Omega^{\cdot}(Z,F)}$  defines an  $L^{2}$ -metric on  $\Omega^{\cdot}(Z,F)$ .

Let  $d^{Z,*}$  be the formal adjoint of  $d^Z$  with respect to  $\langle \, , \rangle_{\Omega^{\cdot}(Z,F)}$ . Put

$$\Box^{Z} = d^{Z}d^{Z,*} + d^{Z,*}d^{Z}. \tag{1.4}$$

Then,  $\square^Z$  acts on  $\Omega^{\cdot}(Z, F)$  and preserves its degree. It is a formally self-adjoint non-negative second-order elliptic differential operator. Since Z is compact,  $\square^Z$  has a unique self-adjoint extension, which we still denote by  $\square^Z$ .

<sup>&</sup>lt;sup>3</sup>We use the convention that  $F = \pi_1(Z) \setminus (X \times \mathbf{C}^r)$ , where  $\pi_1(Z)$  acts on the left on the universal covering X of Z by the deck transformation, and acts on the left on  $\mathbf{C}^r$  by  $\rho$ .

Take  $0 \leq q \leq \dim Z$ . Let  $\square_q^Z$  be the restriction of  $\square^Z$  on  $\Omega^q(Z, F)$ . By the Hodge theory, we have a canonical isomorphism of vector spaces,

$$\ker \Box_q^Z \simeq H^q(Z, F).$$
 (1.5)

Denote by  $(\Box_q^Z)^{-1}$  the inverse of  $\Box_q^Z$  acting on the orthogonal space of  $\ker \Box_q^Z$  in  $\Omega^q(Z,F)$ .

**Definition 1.3.** For  $s \in \mathbb{C}$  such that  $\operatorname{Re}(s) > \dim \mathbb{Z}/2$ , set

$$\theta_q(s) = \text{Tr}\left[\left(\Box_q^Z\right)^{-s}\right].$$
 (1.6)

By [Se67] or [BeGeVe04, Proposition 9.35],  $\theta_q(s)$  has a meromorphic extension to **C** which is holomorphic at s=0. The regularized determinant of  $\Box_q^Z$  is a positive number defined by  $\exp(-\theta_q'(0))$ . We write

$$\det\left(\Box_q^Z\right) = \exp\left(-\theta_q'(0)\right). \tag{1.7}$$

**Definition 1.4.** The Ray-Singer analytic torsion [RS71] of F is defined by

$$T_F(Z) = \prod_{q=1}^{\dim Z} \det \left( \Box_q^Z \right)^{(-1)^q q/2} = \exp \left( \frac{1}{2} \sum_{q=1}^{\dim Z} (-1)^{q-1} q \, \theta_q'(0) \right) \in \mathbf{R}_+^*. \tag{1.8}$$

Let o(TZ) be the orientation line bundle of Z. Let  $\overline{F}^*$  be the antidual bundle of F. The following proposition is a consequence of the Poincaré duality.

**Proposition 1.5.** The following identity holds,

$$T_F(Z) = \left(T_{\overline{F}^* \otimes o(TZ)}(Z)\right)^{(-1)^{\dim Z - 1}}.$$
(1.9)

In particular, if Z has even dimension and is orientable, and if F is unitarily flat, then

$$T_F(Z) = 1. (1.10)$$

The next theorem is a special case of the anomaly formula of Bismut–Zhang [BZ92, Theorem 4.7], which generalizes a result of Ray–Singer [RS71, Theorem 2.1]. Its proof is based on the local index techniques, and consists in calculating the constant term in the asymptotic expansion of  $\text{Tr}[A\exp(-t\Box^Z)]$  as  $t \to 0$ , where A is a certain smooth section of  $\text{End}(\Lambda^{\cdot}(T^*Z) \otimes_{\mathbf{R}} F)$ .

**Theorem 1.6.** Assume  $H^{\cdot}(Z, F) = 0$ . The following statements hold.

- If Z has odd dimension, then  $T_F(Z)$  is independent of  $g^{TZ}, g^F$ .
- If F is unimodular (i.e.,  $|\det \rho(\gamma)| = 1$  for any  $\gamma \in \pi_1(X)$ ), then  $T_F(Z)$  is independent of  $g^{TZ}$  and the unimodular metric  $g^F$  (i.e.,  $g^F$  indues a flat metric on the determinant line bundle  $\det F = \Lambda^{\max} F$ ).

In either of the above situations, the analytic torsion becomes a topological invariant. The celebrated Cheeger–Müller/Bismut–Zhang Theorem [C79, M78, M93, BZ92] compares the Ray–Singer analytic torsion with the Reidemeister combinatorial torsion [Re35, Fr35, dR50]. We refer the reader to the above references for more details.

**Example 1.7.** We use the notation in Example 1.1. We equip TZ with the standard metric  $(dt)^2$  and equip F with the trivial metric. Then,

etric 
$$(dt)^2$$
 and equip  $F$  with the trivial metric. Then,
$$d^Z = dt \frac{\partial}{\partial t} + \alpha dt. \quad d^{Z,*} = -i_{\frac{\partial}{\partial t}} \frac{\partial}{\partial t} + \overline{\alpha} i_{\frac{\partial}{\partial t}}, \quad \Box^Z = -\left(\frac{\partial}{\partial t} - \overline{\alpha}\right) \left(\frac{\partial}{\partial t} + \alpha\right). \tag{1.11}$$

Using the Fourier transform, we get the spectrum

$$\operatorname{Sp} \square_0^Z = \operatorname{Sp} \square_1^Z = \left\{ \left| \alpha + 2ik\pi \right|^2 : k \in \mathbf{Z} \right\}, \tag{1.12}$$

and

$$T_F(Z) = \begin{cases} \left| 2\sinh\left(\frac{\alpha}{2}\right) \right|^{-1} &, & \alpha \notin 2i\pi \mathbf{Z}, \\ 1 &, & \alpha \in 2i\pi \mathbf{Z}. \end{cases}$$
 (1.13)

We also have [BZ92, Theorem 0.3].

**Theorem 1.8.** We use the notation in Example 1.2. Assume that two of  $F_i$  are acyclic. If Z has odd dimension, or if  $g^{F_0}$ ,  $g^{F_1}$ , and  $g^{F_2}$  induce a flat metric on  $\det F_0 \otimes (\det F_1)^{-1} \otimes \det F_2$ , then

$$T_{F_1}(Z) = T_{F_0}(Z)T_{F_2}(Z). (1.14)$$

Here, for a line bundle L, the notion  $L^{-1}$  denotes the dual bundle of L.

### 1.3. A formula for fibrations

Let  $\pi: M \to Z$  be a fibration of closed manifolds with closed fibre Y. Let  $TY \subset TM$  be the relative tangent bundle on M of the fibre Y. Let F be a flat vector bundle on M.

Let  $g^{TM}, g^{TY}, g^{TZ}$  be the Riemannian metrics on TM, TY, TZ. Let  $g^F$  be a Hermitian metric on F. Let  $T_F(M)$  be the Ray–Singer analytic torsion with respect to  $g^{TM}$  and  $g^F$ .

Moreover, the restriction  $F|_Y$  of F to the fibre Y is still a flat vector bundle. So we can consider the fibrewise Ray–Singer analytic torsion  $T_{F|_Y}(Y)$  with respect to  $g^{TY}$  and  $g^F|_Y$ . It is a smooth function on Z.

Also, for  $0 \le q \le \dim Y$ , the fibrewise cohomology  $H^q(Y, F|_Y)$  of Y forms a flat vector bundle on Z (see [BLo95, Section III.f]). Using the Hodge theory (1.5), the  $L^2$  metric on  $\Omega^q(Y, F|_Y)$  induces a Hermitian metric  $g^{H^q(Y, F|_Y)}$  on  $H^q(Y, F|_Y)$ . Let  $T_{H^q(Y, F|_Y)}(Z)$  be the Ray–Singer analytic torsion with respect to  $g^{TZ}$  and  $g^{H^p(Y, F|_Y)}$ .

The following theorem is a special case of a very general result of Ma [Ma02, Theorem 0.1 and (0.6)], which is a generalization of Dai–Melrose [DMe12] and Lück–Schick–Thielmann [LüScTh98, Theorem 0.2].

**Theorem 1.9.** Assume  $H^{\cdot}(M, F) = 0$  and  $H^{\cdot}(Z, H^{\cdot}(Y, F|_{Y})) = 0$ . If M has odd dimension or if F is unimodular, then

$$T_F(M) = \exp\left(\int_Z e\left(TZ, \nabla^{TZ}\right) \log T_{F|Y}(Y)\right) \prod_{q=0}^{\dim Y} \left(T_{H^q(Y, F|Y)}(Z)\right)^{(-1)^q}, (1.15)$$

where  $e\left(TZ, \nabla^{TZ}\right) \in \Omega^{\dim Z}(Z, o(TZ))$  is the Euler characteristic form on Z with respect to the Levi-Civita connection  $\nabla^{TZ}$ . In addition, if Z has odd dimension, then

$$T_F(M) = \prod_{q=0}^{\dim Y} \left( T_{H^q(Y,F|_Y)}(Z) \right)^{(-1)^q}. \tag{1.16}$$

### 2. The Fried conjecture

It was known dating back to Poincaré, Hopf, Lefschetz, etc., that certain topological invariants, like Euler characteristic number, can be expressed with the help of some dynamical objects, like the fixed point set. Fried considered a similar problem relating the Ray–Singer analytic torsion with the closed orbits of certain dynamical systems, which he called the Lefschetz formula for flows.

More precisely, given a diffeomorphism of a manifold, the Lefschetz fixed point formula asserts that the Lefschetz number can be written as a sum of the Lefschetz indices of each fixed point of the diffeomorphism. Formally, the Lefschetz index is a way to count cohomologically the cardinality of the fixed point set.

For a dynamical flow, Fried uses the Fuller index [Fu67] to count cohomologically the cardinality of the set of closed orbits of the flow. However, due to the infinite numbers of closed orbits, the Fuller index of the set of all closed orbits is not well defined and should be regularized. For this Fried relies on the meromorphic extension and the regularity at 0 of the Ruelle dynamical zeta function. The purpose of this section is to formulate the above as a general conjecture, known as the Fried conjecture.

This section is organized as follows. In Section 2.1, we consider a diffeomorphism on a manifold. We recall the Lefschetz fixed point formula.

In Section 2.2, we consider a flow on a manifold. We introduce the Fuller index.

In Section 2.3, we construct the Ruelle dynamical zeta function, and state the Fried conjecture.

#### 2.1. Lefschetz index

Let Z be a closed manifold. Let  $T \in \text{Diffeo}(Z)$  be a diffeomorphism of Z. The pull back  $T^*$  defines a morphism of the complex  $(\Omega^{\cdot}(Z), d)$ . It induces a morphism on

the cohomology  $H^{\cdot}(Z)$ . The Lefschetz number of T is defined by

$$L_T = \text{Tr}_s^{H^*(Z)}[T^*] = \sum_{q=0}^{\dim Z} (-1)^q \text{Tr}\left[T^*|_{H^q(Z)}\right].$$
 (2.1)

Clearly, if T is the identity map, then the Lefschetz number  $L_T$  is equal to the Euler characteristic number  $\chi(Z)$ .

Let

$$Z^{T} = \{ z \in Z : Tz = z \} \tag{2.2}$$

be the set of the fixed points of T. Recall that a fixed point  $z \in Z^T$  is called non-degenerate if  $\det(1 - D_z T)|_{T_z Z} \neq 0$ . For such a fixed point  $z \in Z^T$ , the Lefschetz index of T at z is defined by

$$\operatorname{ind}_{L}(T,z) = \operatorname{sgn}(\det(1 - D_{z}T)|_{T_{z}Z}). \tag{2.3}$$

If T has only non-degenerate fixed points, then  $Z^T$  is finite, and the Lefschetz fixed point theorem [BeGeVe04, Corollary 6.7] states that

$$L_T = \sum_{z \in Z^T} \operatorname{ind}_L(T, z). \tag{2.4}$$

More generally, if  $\mathbb{Z}^T$  is not finite, we need the following assumption.

### Assumption 2.1. We assume that

(1) the fixed point set  $Z^T$  is a disjoint union of finitely many closed connected submanifolds  $Z_i \subset Z$ , so that

$$Z^T = \coprod_{i \in I} Z_i; \tag{2.5}$$

(2) for  $z \in Z^T$ , the eigenspace of  $D_zT : T_zZ \to T_zZ$  associated to the eigenvalue 1 coincides with its characteristic space, and is equal to  $T_zZ^T$ .

Let  $N_{Z_i/Z} = (TZ|_{Z_i})/TZ_i$  be the normal bundle of  $Z_i$  in Z. Our assumption (2) is equivalent to

$$\det(1 - D_z T)|_{N_{Z_i/Z_i,z}} \neq 0, \qquad \text{for all } i \in I, z \in Z_i.$$
 (2.6)

Note that for each  $i \in I$ , the sign of the above determinant does not depend on  $z \in Z_i$ .

**Definition 2.2.** The Lefschetz index of the diffeomorphism T at  $Z_i$  is defined by

$$\operatorname{ind}_{L}(T, Z_{i}) = \operatorname{sgn}\left(\det(1 - D_{z}T)|_{N_{Z_{i}/Z, z}}\right) \cdot \chi(Z_{i}) \in \mathbf{Z},\tag{2.7}$$

where  $z \in Z_i$ .

Under Assumption 2.1, we have the Lefschetz fixed point theorem<sup>4</sup>

$$L_T = \sum_{i \in I} \operatorname{ind}_L(T, Z_i). \tag{2.8}$$

Formally, Lefschetz index  $\operatorname{ind}_L(T, Z_i)$  counts cohomologically the cardinality of  $Z_i$ . Equation (2.8) says that the Lefschetz number can be obtained by counting cohomologically the cardinality of the fixed point set  $Z^T$ .

#### 2.2. Fuller index

Let M be a closed manifold. Let  $V \in C^{\infty}(M, TM)$  be a smooth vector field on M, and let  $(\phi_t)_{t \in \mathbf{R}}$  be the flow generated by V.

**Definition 2.3.** We define the periodic set

$$\wp(\phi_{\cdot}) = \{ (x, t) \in M \times \mathbf{R}_{+}^{*} : \phi_{t}(x) = x \}, \tag{2.9}$$

and the length spectrum

$$\ell(\phi_{\cdot}) = \{ t \in \mathbf{R}_{+}^{*} : \text{ there is } x \in M \text{ such that } \phi_{t}(x) = x \}.$$
 (2.10)

A point  $(x,t) \in \wp(\phi)$  corresponds to a closed orbit  $\{\phi_s(x)\}_{0 \leq s \leq t}$  starting from x of period t. And  $\ell(\phi)$  represents all the possible periods. Clearly, if  $\ell \in \ell(\phi)$ , then  $\ell \in \ell(\phi)$  for all  $\ell \in \mathbb{N}^*$ .

Since our ultimate object "closed orbit" disregards the starting points, we will consider certain quotient space of  $\wp(\phi)$ . Define an **R**-action on  $\wp(\phi)$  by

$$s \cdot (x,t) = (\phi_{ts}x,t),$$
 for  $s \in \mathbf{R}, (x,t) \in \wp(\phi_{\cdot}).$  (2.11)

By (2.11), the group  $\mathbb{S}^1 = \mathbf{R}/\mathbf{Z}$  acts on  $\wp(\phi)$ . Set

$$\overline{\wp}(\phi_{\cdot}) = \wp(\phi_{\cdot})/\mathbb{S}^{1}. \tag{2.12}$$

We identify  $\overline{\wp}(\phi)$  with the space of closed orbits of the flow  $\phi$ ..

To count cohomologically the points in the  $\overline{\wp}(\phi)$ , as in Section 2.1, we need an analogue of Assumption 2.1 for flows.

### Assumption 2.4. We assume that

- (1) the length spectrum  $\ell(\phi) \subset (0, \infty)$  is discrete;
- (2) for any  $\ell \in \ell(\phi)$ , the diffeomorphism  $\phi_{\ell} \in \text{Diffeo}(M)$  satisfies Assumption 2.1.

Recall that  $M^{\phi_{\ell}} \subset M$  is the fixed point set of  $\phi_{\ell}$  in M. By (1) of Assumption 2.4, we can write

$$\wp(\phi.) = \coprod_{\ell \in \ell(\phi.)} M^{\phi_{\ell}} \times \{\ell\}. \tag{2.13}$$

<sup>&</sup>lt;sup>4</sup>When T is an isometry of Z, a proof can be found in [BeGeVe04, Section 6.4]. For general T, the Lefschetz fixed point formula can be proved by adapting the argument given in [BZ92, Theorem 4.20].

To avoid the problem of disconnectedness of  $M^{\phi_{\ell}}$ , we write instead

$$\wp(\phi_{\cdot}) = \coprod_{i \in I} M_i \times \{\ell_i\}, \tag{2.14}$$

where  $M_i$  is a connected component of  $M^{\phi_{\ell_i}}$ . Note that not all the  $\ell_i$  are necessarily distinct. Moreover, (1) of Assumption 2.4 implies that the  $\mathbb{S}^1$ -action on  $\wp(\phi_i)$  is locally free. Thus,  $M_i/\mathbb{S}^1$  is a closed connected orbifold. By (2.12), we have

$$\overline{\wp}(\phi_{\cdot}) = \prod_{i \in I} M_i / \mathbb{S}^1 \times \{\ell_i\}. \tag{2.15}$$

Denote by  $\chi_{\text{orb}}(M_i/\mathbb{S}^1) \in \mathbf{Q}$  the orbifold Euler characteristic number [Sa57] of  $M_i/\mathbb{S}^1$ . Set

$$m_i = \left| \ker \left( \mathbb{S}^1 \to \text{Diffeo}(M_i) \right) \right| \in \mathbf{N}^*$$
 (2.16)

to be the generic multiplicity of the closed orbits in  $M_i/\mathbb{S}^1 \times \{\ell_i\}$ .

By (2) of Assumption 2.4, as in Section 2.1, the sign

$$\operatorname{sgn}\left(\det(1 - D\phi_{\ell_i})|_{N_{M_i/M}}\right) \in \{\pm 1\}$$
(2.17)

is well defined for all  $i \in I$ .

**Definition 2.5.** The Fuller index of the flow  $\phi$  at  $M_i \times \{\ell_i\}$  is defined by

$$\operatorname{ind}_{F}(\phi_{\cdot}, M_{i}) = \operatorname{sgn}\left(\det(1 - D\phi_{\ell_{i}})|_{N_{M_{i}/M}}\right) \cdot \frac{\chi_{\operatorname{orb}}(M_{i}/\mathbb{S}^{1})}{m_{i}} \in \mathbf{Q}.$$
 (2.18)

Remark 2.6. The pair  $(M_i, \mathbb{S}^1)$  defines a non-effective orbifold. Its "non-effective" Euler characteristic number is given by  $\frac{\chi_{\text{orb}}(M_i/\mathbb{S}^1)}{m_i}$ . In this way, the Fuller index  $\text{ind}_F(\phi_i, M_i)$  is a strict analogue of the Lefschetz index (2.7).

More generally, let F be a flat vector bundle on M with holonomy  $\rho$ . Since  $M_i$  is connected, for any  $x \in M_i$ , the closed orbit  $\{\phi_t(x)\}_{0 \le t \le \ell_i}$  lies in the same freely homotopy class of the loops on M. Up to conjugation, the holomony<sup>5</sup>  $\rho_i$  of F associated to the closed orbit  $\{\phi_t(x)\}_{0 \le t \le \ell_i}$  does not depend on the choice of  $x \in M_i$ . We can define the twisted Fuller index by

$$\operatorname{ind}_{F}(\phi_{\cdot}, M_{i}, F) = \operatorname{ind}_{F}(\phi_{\cdot}, M_{i}) \operatorname{Tr} \left[\rho_{i}\right]. \tag{2.19}$$

### 2.3. The Ruelle dynamical zeta function

We assume that the flow satisfies Assumption 2.4. As an analogue of (2.4), Fried raised the question: for what kind of flows the infinite sum

$$\sum_{i \in I} \operatorname{ind}_F(\phi_{\cdot}, M_i, F) \tag{2.20}$$

<sup>&</sup>lt;sup>5</sup>The morphism  $\rho_i$  can be obtained by the parallel transport with respect to the flat connection along the closed orbit  $\{\phi_t(x)\}_{0 \leqslant t \leqslant \ell_i}$  from  $t = \ell_i$  to t = 0.

can be regularized, and is equal to some topological invariant. More precisely, Fried conjectured that if F is unitarily flat and acyclic, then the regularized sum (2.20) is equal to  $\log T_F(M)$ .

To regularize the sum (2.20), Fried used the Ruelle dynamical zeta function [Ru76b, F87].

**Definition 2.7.** For  $\sigma \in \mathbb{C}$ , we define formally

$$R_{\phi,\rho}(\sigma) = \exp\left(\sum_{i \in I} \operatorname{ind}_F(\phi_i, M_i, F) e^{-\ell_i \sigma}\right). \tag{2.21}$$

We call  $R_{\phi,\rho}$  is well defined if the following properties hold.

- 1. There is  $\sigma_0 > 0$  such that for  $\sigma \in \mathbf{C}$  with  $\operatorname{Re}(\sigma) > \sigma_0$ , the sum on the right-hand side of (2.21) converges to a holomorphic function;
- 2. The holomorphic function  $R_{\phi,\rho}(\sigma)$ , defined for Re  $(\sigma) > \sigma_0$ , has a meromorphic extension to  $\sigma \in \mathbf{C}$ .

**Remark 2.8.** The above definition of the Ruelle dynamical zeta function is the reciprocal of the one introduced by Fried [F87, Section 5].

The Fried conjecture now can be formulated as follows.

Conjecture 2.9. For a wide class of flows on a closed manifold M, if F is a unitarily flat vector bundle on M and  $H^{\cdot}(M,F)=0$ , then the Ruelle dynamical zeta function  $R_{\phi,\rho}(\sigma)$  is well defined, and is regular at  $\sigma=0$  such that

$$|R_{\phi,\rho}(0)| = T_F(M).$$
 (2.22)

**Example 2.10.** Consider the rotation flow  $\phi$ . on  $\mathbb{S}^1$ , i.e.,

$$\phi_t(x) = x + t \mod \mathbf{Z}, \qquad \text{for } t \in \mathbf{R}, x \in \mathbf{R}/\mathbf{Z}.$$
 (2.23)

For the flat vector bundle defined in Example 1.1, the Ruelle dynamical zeta function is given by

$$R_{\phi,\rho}(\sigma) = (1 - e^{-\sigma + \alpha})^{-1}.$$
 (2.24)

If F is acyclic and unitarily flat (i.e.,  $\alpha \in i\mathbf{R}$  and  $\alpha \notin 2i\pi\mathbf{Z}$ ), by (1.13) and (2.24), we have

$$|R_{\phi,\rho}(0)| = T_F\left(\mathbb{S}^1\right). \tag{2.25}$$

**Example 2.11.** Consider now the geodesic flow of  $\mathbb{S}^1$ . That is a flow  $\widetilde{\phi}$  defined on  $\mathbb{S}^1 \times \{\pm 1\}$ . On  $\mathbb{S}^1 \times \{1\}$ ,  $\widetilde{\phi}$  is just the rotation flow considered in Example 2.10, and on the other copy  $\mathbb{S}^1 \times \{-1\}$ ,  $\widetilde{\phi}$  is the inverse of the rotation flow. Let  $\pi: \mathbb{S}^1 \times \{\pm 1\} \to \mathbb{S}^1$  be the natural projection.

Recall that F is defined in Example 1.1. Let  $\pi^*F$  be the pull back of the flat vector bundle on  $\mathbb{S}^1 \times \{\pm 1\}$ . The Ruelle dynamical zeta function is then given by

$$R_{\widetilde{\phi},\rho}(\sigma) = (1 - e^{-\sigma + \alpha})^{-1} (1 - e^{-\sigma - \alpha})^{-1}.$$
 (2.26)

If  $\alpha \in \mathbf{C}$  and  $\alpha \notin 2i\pi \mathbf{Z}$ , we have

$$|R_{\widetilde{\phi},\rho}(0)| = T_{\pi^*F} \left( \mathbb{S}^1 \times \{\pm 1\} \right).$$
 (2.27)

Example 2.10 shows that the Fried conjecture fails for non-unitarily flat vector bundles, while Example 2.11 shows that for some special flow the Fried conjecture holds even for certain non-unitarily flat vector bundles.

However, even for unitarily flat vector bundles, we can not expect that the Fried conjecture holds for general flows. In fact, a fixed point or a closed orbit is a limit set of dimensions 0 or 1. Wilson [Wi66, Corollary 2] has constructed a smooth flow on any closed manifold M with vanishing Euler characteristic number, whose only limit sets are a finite collection of torus of dimension dim M-2. In particular, if dim  $M \ge 4$  with  $\chi(M) = 0$ , there exists a flow without fixed points or closed orbits. If dim M = 3, Schweitzer [Sch74, Theorem A] has constructed a  $C^1$ -flow without fixed points or closed orbits.

In the following sections, we will discuss several different classes of flows where the Fried conjecture has or might have a positive solution. This includes the suspension flow, the Morse–Smale flow, the geodesic flow, and the Anosov flow.

### 3. Suspension flow of a diffeomorphism

In this section, we consider the suspension flow of a diffeomorphism. We show that in this case the Fried conjecture is a consequence of the Lefschetz fixed point formula (2.8).

This section is organized as follows. In Section 3.1, we introduce the suspension flow.

In Section 3.2, we describe the flat vector bundle on a suspension, and evaluate its analytic torsion.

In Section 3.3, we give an explicit formula for the Ruelle dynamical zeta function associated to the suspension flow. Then, we show the Fried conjecture.

#### 3.1. The definition of the suspension flow

Let Y be a connected closed manifold. Let  $T \in \text{Diffeo}(Y)$  be a diffeomorphism of Y. Let M be the suspension of T. Then  $M = \mathbf{R} \times_{\mathbf{Z}} Y$  is the quotient space of  $\mathbf{R} \times Y$  by the **Z**-action given by

$$n \cdot (t, y) = (t - n, T^n y),$$
 for  $n \in \mathbf{Z}, (t, y) \in \mathbf{R} \times Y.$  (3.1)

Clearly,  $M \to \mathbf{R}/\mathbf{Z}$  is a fibration with fibre Y.

The suspension flow  $\phi$  on M is defined by

$$\phi_s([t, y]) = [t + s, y], \qquad \text{for } s \in \mathbf{R}, [t, y] \in \mathbf{R} \times_{\mathbf{Z}} Y. \tag{3.2}$$

### 3.2. The flat vector bundle on the suspension

Let E be a flat vector bundle on Y. Assume that the diffeomorphism T lifts to E. This means that there is a bundle morphism  $T_E$  of E such that the diagram

$$E \xrightarrow{T_E} E$$

$$\downarrow \qquad \qquad \downarrow$$

$$Y \xrightarrow{T} Y$$

$$(3.3)$$

commutes. Let

$$F = \mathbf{R} \times_{\mathbf{Z}} E \tag{3.4}$$

be the suspension of E with respect to  $T_E$ . Then F is a flat vector bundle on M.

**Proposition 3.1.** All the flat vector bundles on M can be obtained in this way.

*Proof.* Using the long exact sequence of the homotopy groups associated to the fibration  $M \to \mathbf{R}/\mathbf{Z}$ , we get an exact sequence of groups

$$1 \to \pi_1(Y) \to \pi_1(M) \to \mathbf{Z} \to 1. \tag{3.5}$$

The diffeomorphism T defines an isomorphism  $T_*: \pi_1(Y) \to \pi_1(Y)$  of groups such that  $\pi_1(M)$  is the semidirect product of  $\pi_1(Y)$  and **Z** with respect to  $T_*$ .

Any r-dimensional representation of  $\pi_1(M)$  can be constructed by an r-dimensional representation  $\rho$  of  $\pi_1(Y)$  and an element  $A \in GL_r(\mathbb{C})$  such that

$$A\rho(\gamma) = \rho(T_*\gamma)A,$$
 for all  $\gamma \in \pi_1(Y).$  (3.6)

It is easy to check that  $\rho$  and A induce our E and  $T_E$ .

As in Section 2.1, the pull back  $T_E^*$  defines a morphism on  $H^{\cdot}(Y, E)$ . Let  $H_T^{\cdot}(Y, E)$  be a flat vector bundle on  $\mathbf{R}/\mathbf{Z}$  defined by the holonomy

$$\rho: n \in \mathbf{Z} \to \left(T_E^*\right)^n. \tag{3.7}$$

This is just the direct image  $H'(Y, F|_Y)$  of F described in Section 1.3.

**Proposition 3.2.** The flat vector bundle F on M is acyclic if and only if 1 is not an eigenvalue of  $T_E^*$  on  $H^{\cdot}(Y, E)$ .

*Proof.* Let  $(E_r^{p,q}, d_r)_{r\geqslant 0}$  be the Leray spectral sequence [BoTu82, p. 169] associated to the fibration  $M\to \mathbf{R}/\mathbf{Z}$ . By [BoTu82, Theorem 14.18],  $(E_r^{p,q}, d_r)_{r\geqslant 0}$  converges to  $H^{\cdot}(M, F)$ , and

$$E_2^{p,q} = H^p(\mathbf{R}/\mathbf{Z}, H_T^q(Y, E)).$$
 (3.8)

By (3.8), for  $r \ge 2$ ,  $p \ne 0, 1$ , we have  $E_r^{p,q} = 0$ . Since  $d_r$  sends  $E_r^{p,q}$  to  $E_r^{p+r,q-r+1}$ , we see that  $d_2 = d_3 = \cdots = 0$ . So the spectral sequence degenerates at r = 2. It implies

$$\dim H^{p}(M,F) = \dim H^{0}(\mathbf{R}/\mathbf{Z}, H_{T}^{p}(Y,E)) + \dim H^{1}(\mathbf{R}/\mathbf{Z}, H_{T}^{p-1}(Y,E)).$$
(3.9)

By (3.9),  $H^{\cdot}(M, F) = 0$  if and only if  $H^{\cdot}(\mathbf{R}/\mathbf{Z}, H_T^{\cdot}(Y, E)) = 0$ . By Example 1.1, we see that our proposition holds true if  $T_E^*$  is diagonalizable. Thanks to Example 1.2, we can deduce our proposition in full generality.

If A is a matrix acting on a **Z**-graded space E which preserves the degree, then write

$$\det_{\mathbf{s}}(A)|_{E^{\cdot}} = \prod_{q \in \mathbf{Z}} \left( \det(A)|_{E^q} \right)^{(-1)^q}. \tag{3.10}$$

**Proposition 3.3.** Assume  $H^{\cdot}(M, F) = 0$ . If M has odd dimension or if F is unimodular, then

$$T_F(M) = \left| \det_{\mathbf{s}} (T_E^*) \left|_{H^{\cdot}(Y,E)} \right|^{1/2} \left| \det_{\mathbf{s}} (1 - T_E^*) \left|_{H^{\cdot}(Y,E)} \right|^{-1}.$$
 (3.11)

*Proof.* By Theorem 1.9, we have

$$T_F(M) = \prod_{q=0}^{\dim Y} \left( T_{H_T^q(Y,E)} (\mathbf{R}/\mathbf{Z}) \right)^{(-1)^q}.$$
 (3.12)

Now (3.11) is a consequence of (1.13), (1.14), and (3.12).

### 3.3. The Ruelle dynamical zeta function of the suspension flow

Note that if for all  $n \in \mathbb{N}^*$ , the diffeomorphism  $T^n$  satisfies Assumption 2.1, then the suspension flow  $\phi$ , associated to T satisfies Assumption 2.4.

**Theorem 3.4.** Assume that for all  $n \in \mathbb{N}^*$ , the diffeomorphism  $T^n \in \text{Diffeo}(Y)$  satisfies Assumption 2.1. The Ruelle dynamical zeta function is given by

$$R_{\phi,\rho}(\sigma) = \left(\det_{\mathbf{s}} \left(1 - T_E^* e^{-\sigma}\right) |_{H^{\cdot}(Y,E)}\right)^{-1}.$$
 (3.13)

If  $H^{\cdot}(M,F)=0$ , then  $R_{\phi,\rho}$  is regular at  $\sigma=0$ , so that

$$R_{\phi,\rho}(0) = \left(\det_{\mathbf{s}} (1 - T_E^*) |_{H^{\perp}(Y,E)}\right)^{-1}.$$
 (3.14)

*Proof.* For the suspension flow, we have  $\ell(\phi) \subset \mathbf{N}^*$ . The periodic set is given by

$$\wp(\phi_{\cdot}) = \prod_{n=1}^{\infty} M^{\phi_n} \times \{n\}. \tag{3.15}$$

Let  $\pi: \mathbf{R} \times Y \to M$  be the natural projection. We have

$$M^{\phi_n} = \pi \left( [0, 1] \times Y^{T^n} \right). \tag{3.16}$$

We claim that Fuller index (twisted by F) of  $\phi$ . at  $M^{\phi_n}$  is given by

$$\operatorname{ind}_{F}(\phi_{\cdot}, M^{\phi_{n}}, F) = \frac{1}{n} \operatorname{Tr}_{s}^{H^{\cdot}(Y, E)} [(T_{E}^{*})^{n}].$$
 (3.17)

Let us show (3.17) in the case where  $Y^{T^n}$  is discrete. The proof for the general case is similar, and we omit the details. For  $p \in \mathbf{N}^*$  and p|n, let  $Y_p^{T^n} \subset Y^{T^n}$  be

the subset of  $Y^{T^n}$  formed by the points of  $Y^{T^n}$  whose prime period is p. Take  $y \in Y_p^{T^n}$ . By (2.18), the contribution of the closed orbit  $\{[t,y]\}_{0 \leqslant t \leqslant n}$  to the Fuller index (twisted by F) is given by

$$\frac{\operatorname{sgn}\left(\operatorname{det}\left(1-DT_{y}^{n}\right)|_{T_{y}Y}\right)}{n/p}\operatorname{Tr}\left[\tau_{y,n}\right],\tag{3.18}$$

where  $\tau_{y,n}$  is the parallel transport of F with respect to the flat connection along the curve  $\{[t,y]\}_{0 \le t \le n}$  from t=n to t=0. We have

$$\operatorname{Tr}\left[\tau_{y,n}\right] = \operatorname{Tr}^{E_y}\left[T_E^{-n}\right]. \tag{3.19}$$

Since  $y, Ty, \dots, T^{p-1}y$  are all in  $Y_p^{T^n}$  and correspond to the same closed orbit, we have

$$\operatorname{ind}_{F}(\phi_{\cdot}, M^{\phi_{n}}, F) = \sum_{p|n} \frac{1}{p} \sum_{y \in Y_{p}^{T^{n}}} \frac{\operatorname{sgn}\left(\det\left(1 - DT_{y}^{n}\right)|_{T_{y}Y}\right)}{n/p} \operatorname{Tr}^{E_{y}}\left[T_{E}^{-n}\right]$$

$$= \frac{1}{n} \sum_{y \in Y^{T^{n}}} \operatorname{sgn}\left(\det\left(1 - DT_{y}^{n}\right)|_{T_{y}Y}\right) \operatorname{Tr}^{E_{y}}\left[T_{E}^{-n}\right]. \quad (3.20)$$

By a twisted version of (2.4) (see [BeGeVe04, Theorem 6.6]) and (3.20), we get (3.17).

By (3.17), using

$$\log(1-z) = -\sum_{n=1}^{\infty} \frac{z^n}{n},$$
(3.21)

we get (3.13). The rest of our proposition is a consequence of Proposition 3.2 and (3.13).

**Corollary 3.5.** Suppose that for all  $n \in \mathbb{N}^*$  the diffeomorphism  $T^n \in \text{Diffeo}(Y)$  satisfies Assumption 2.1. Assume that  $H^{\cdot}(M, F) = 0$  and that F is unimodular. Then, we have

$$|R_{\phi,\rho}(0)| = T_F(M).$$
 (3.22)

*Proof.* By (3.11) and (3.14), it is enough to show

$$\left| \det_{\mathbf{s}}(T_E^*) |_{H^{\cdot}(Y,E)} \right| = 1.$$
 (3.23)

Recall that  $H_T(Y, E)$  is a flat vector bundle on  $\mathbf{R}/\mathbf{Z}$ . Equation (3.23) is equivalent to say that the line bundle  $\bigotimes_{q=0}^{\dim Y} \left(\det H_T^q(Y, E)\right)^{(-1)^q}$  is unitarily flat.

Remark that (see [BLo95, Section I.g]) if  $(W, \nabla^W)$  is a flat vector bundle on a manifold S, for any Hermitian metric  $g^W$  on W, the 1-form  $\frac{1}{2}\operatorname{Tr}\left[g^{W,-1}\nabla^Wg^W\right]$  is closed. Moreover, its class in  $H^1(S)$  does not depend on the choice of  $g^W$ , and is called the first odd Chern class  $c_1^{\operatorname{odd}}(W)$  of W. Clearly,  $\det(W)$  is unitarily flat if and only if  $c_1^{\operatorname{odd}}(W) = 0$ .

By above, it is equivalent to show the first odd Chern class of the fibrewise cohomology vanishes,

$$\sum_{q=0}^{\dim Y} (-1)^q c_1^{\text{odd}}(H_T^q(Y, E)) = 0.$$
 (3.24)

Since F is unimodular, (3.24) is a consequence of [BLo95, Theorem 0.1].

### 4. Morse-Smale flow

The Morse–Smale flow is the simplest structurally stable dynamical system which has only two types of recurrent behaviors: closed orbits and fixed points. In this section, we study the Fried conjecture for the Morse–Smale flow. We show that the Fried conjecture in this case is a consequence of Cheeger–Müller/Bismut–Zhang theorem. The result in this section is originally due to Fried [F87, Section 3] and is extended by Shen–Yu [ShY18].

This section is organized as follows. In Section 4.1, we introduce the Morse–Smale flow.

In Section 4.2, we give an explicit formula for the Ruelle dynamical zeta function, and show the Fried conjecture.

### 4.1. The definition of Morse-Smale flow

Let M be a closed manifold with a smooth vector field  $V \in C^{\infty}(M, TM)$ . Let  $(\phi_t)_{t \in \mathbf{R}}$  be the flow on M generated by V.

**Definition 4.1.** The nonwandering set of  $\phi$  is defined by

$$\left\{x \in M : \ \forall \text{ open neighborhood } U \text{ of } x, \ \forall T > 0, \text{ we have } U \cap \bigcup_{t \geqslant T} \phi_t(U) \neq \varnothing \right\}.$$

$$(4.1)$$

**Definition 4.2.** A fixed point  $x \in M$  of the flow  $\phi$  is called hyperbolic if there is a  $\phi_t$ -invariant splitting

$$T_x M = E_x^u \oplus E_x^s, \tag{4.2}$$

and there exist  $C > 0, \theta > 0$  and a Riemannian metric on M such that for  $v \in E_x^u$ ,  $v' \in E_x^s$ , and t > 0, we have

$$|\phi_{-t,*}v| \le Ce^{-\theta|t|}|v|, \qquad |\phi_{t,*}v'| \le Ce^{-\theta|t|}|v'|.$$
 (4.3)

The index  $\operatorname{ind}(x) \in \mathbf{N}$  of x is defined by

$$\operatorname{ind}(x) = \dim E_x^u. \tag{4.4}$$

The unstable and stable manifolds of x are defined by

$$W_x^u = \left\{ y \in M : \lim_{t \to -\infty} d_M(\phi_t(y), x) = 0 \right\},\$$

$$W_x^s = \left\{ y \in M : \lim_{t \to +\infty} d_M(\phi_t(y), x) = 0 \right\},$$
(4.5)

where  $d_M$  is the Riemannian distance on M.

**Definition 4.3.** A closed orbit  $\gamma$  of the flow  $\phi$ . is called hyperbolic, if there is a  $\phi_t$ -invariant continuous splitting

$$TM|_{\gamma} = \mathbf{R}V \oplus E_{\gamma}^{u} \oplus E_{\gamma}^{s} \tag{4.6}$$

of vector bundles over  $\gamma$  such that (4.3) holds. The index  $\operatorname{ind}(\gamma) \in \mathbf{N}$  of  $\gamma$  is defined by

$$\operatorname{ind}(\gamma) = \dim E_{\gamma}^{u}. \tag{4.7}$$

The unstable and stable manifolds of  $\gamma$  are defined by

$$W_{\gamma}^{u} = \bigcup_{x \in \gamma} \left\{ y \in M : \lim_{t \to -\infty} d_{M}(\phi_{t}(y), \phi_{t}(x)) = 0 \right\},$$

$$W_{\gamma}^{s} = \bigcup_{x \in \gamma} \left\{ y \in M : \lim_{t \to +\infty} d_{M}(\phi_{t}(y), \phi_{t}(x)) = 0 \right\}.$$

$$(4.8)$$

Denote by A the set of fixed points and by B the set of prime closed orbits.

**Definition 4.4.** A vector field V or a flow  $\phi$ , is called Morse–Smale (see [PalMe82, Definition, p.118]) if

- the sets A and B are finite and contain only hyperbolic elements;
- the nonwandering set of  $\phi$  is equal to  $A \cup \bigcup_{\gamma \in B} \gamma$ ;
- the stable manifold of any element in  $A \coprod B$  intersects transversally with the unstable manifolds of any element in  $A \coprod B$ .

In the rest of this section, we assume that V is a Morse–Smale vector field.

### 4.2. The Ruelle dynamical zeta function of the Morse-Smale flow

For  $\gamma \in B$ , denote by  $\ell_{\gamma} \in \mathbf{R}_{+}^{*}$  its period. A closed orbit  $\gamma \in B$  is called untwist if  $E_{\gamma}^{u}$  is orientable along  $\gamma$ , and is called twist otherwise. Put

$$\Delta(\gamma) = \begin{cases} 1, & \text{if } \gamma \text{ is untwist,} \\ -1, & \text{if } \gamma \text{ is twist.} \end{cases}$$
 (4.9)

Recall that F is a flat vector bundle on M with holonomy  $\rho$ . For  $\gamma \in B$ , denote by  $\rho(\gamma)$  the holonomy along  $\gamma$ , which is also the parallel transport with respect to the flat connection along  $\gamma^{-1}$  (cf. footnote 3). Clearly,  $\rho(\gamma)$  is well defined up to a conjugation.

**Proposition 4.5.** The Ruelle dynamical zeta function is given by

$$R_{\phi,\rho}(\sigma) = \prod_{\gamma \in B} \det \left( 1 - \Delta(\gamma)\rho(\gamma)e^{-\sigma\ell_{\gamma}} \right)^{(-1)^{1+\operatorname{ind}(\gamma)}}.$$
 (4.10)

*Proof.* Without loss of generality, we assume that there is only one prime closed orbit  $\gamma$ . For  $x \in \gamma$ ,  $k \in \mathbf{N}^*$ , let us calculate the sign of  $\det(1 - D\phi_{k\ell_{\gamma}})|_{T_xM/\mathbf{R}V(x)}$ . By (4.6), we have

$$\det(1 - D\phi_{k\ell\gamma})|_{T_xM/\mathbf{R}V(x)} = \det\left(1 - (D\phi_{\ell\gamma})^k\right)|_{E^s_{\gamma,x}} \det\left(1 - (D\phi_{\ell\gamma})^k\right)|_{E^u_{\gamma,x}}.$$
(4.11)

Note that the non-real eigenvalues  $\alpha$  and  $\overline{\alpha}$  of  $(D\phi_{\ell_{\gamma}})^k$  come in pair. Note also that by property of  $E^s$ , all the absolute value of the real eigenvalues of  $(D\phi_{\ell_{\gamma}})^k$  on  $E^s_{\gamma,x}$  is bounded by 1. So the sign of (4.11) comes from the positive eigenvalues of  $D\phi_{\ell_{\gamma}}|_{E^s_{\alpha,x}}$ . More precisely, we have

$$\operatorname{sgn}\left(\det\left(1-\left(D\phi_{\ell_{\gamma}}\right)^{k}\right)|_{T_{x}M/\mathbf{R}V(x)}\right) = \operatorname{sgn}\det\left(-\left(D\phi_{\ell_{\gamma}}\right)^{k}|_{E_{\gamma,x}^{u}}\right)$$
$$= (-1)^{\operatorname{ind}(\gamma)}\Delta(\gamma)^{k}.$$

$$(4.12)$$

Equation 
$$(4.10)$$
 is a consequence of  $(2.21)$ ,  $(3.21)$ , and  $(4.12)$ .

The following theorem is due to Fried [F87, Theorem 3.1] if F is unitarily flat, and can be extended to unimodular vector bundle [ShY18, Proposition 2.12].

**Theorem 4.6.** Assume that F is a unimodular flat vector bundle, and that  $\phi$ . does not have fixed points. If for all  $\gamma \in B$ ,  $\Delta(\gamma)$  is not an eigenvalue of  $\rho(\gamma)$ , then we have  $H^{\cdot}(M,F) = 0$ , and  $R_{\phi,\rho}$  is regular at  $\sigma = 0$ , so that

$$|R_{\phi,\rho}(0)| = \prod_{\gamma \in B} \left| \det \left( 1 - \Delta(\gamma)\rho(\gamma) \right) \right|^{(-1)^{1+\operatorname{ind}(\gamma)}} = T_F(M). \tag{4.13}$$

*Proof.* Following [Fra82, Definition 9.10], let

$$\emptyset = M^0 \subset M^1 \subset \dots \subset M^N = M \tag{4.14}$$

be a Smale filtration on M associated to the flow  $\phi$ . Note that each  $M^p \subset M$  is a submanifold with boundary, and can be constructed by the sublevel set of a smooth Lyapunov function. Also, we have

- on each  $\partial M^p$ , V points toward the inside of  $M^p$ ;
- there is only one prime closed orbit  $\gamma$  in  $M^{p+1}\backslash M^p$  and

$$\gamma = \bigcap_{t \in \mathbf{R}} \phi_t \left( M^{p+1} \backslash M^p \right). \tag{4.15}$$

Note that  $M^{p+1}$  can be obtained from  $M^p$  by attaching a round ind $(\gamma)$ -handle,

$$M^{p+1} = M^p \cup_{\mathbb{S}^1 \times \mathbb{S}^{\operatorname{ind}(\gamma)} \times \mathbb{D}^{m-\operatorname{ind}(\gamma)-1}} \mathbb{S}^1 \times \mathbb{D}^{\operatorname{ind}(\gamma)} \times \mathbb{D}^{m-\operatorname{ind}(\gamma)-1}, \tag{4.16}$$

where  $m = \dim M$ . By [Fra82, Theorem 9.11] (see also [SM96b, Section 2]),

$$H^{q}\left(M^{p+1}, M^{p}, F\right) = H^{q-\operatorname{ind}(\gamma)}\left(\gamma, o(E_{\gamma}^{u}) \otimes_{\mathbf{R}} F|_{\gamma}\right), \tag{4.17}$$

where  $o(E^u_{\gamma})$  is the orientation line bundle of  $E^u_{\gamma}$  along the closed orbit  $\gamma$ .

By Examples 1.1 and 1.2, we see that  $\Delta(\gamma)$  is not an eigenvalue of  $\rho(\gamma)$  if and only if

$$H^{\cdot}(M^{p+1}, M^p, F) = 0.$$
 (4.18)

Using the long exact sequence

$$\cdots \to H^q(M^p, M^{p-1}, F) \to H^q(M^p, F) \to H^q(M^{p-1}, F) \to \cdots,$$
 (4.19)

we see that

$$H^{\cdot}(M,F) = H^{\cdot}(M^{N},F) \simeq H^{\cdot}(M^{N-1},F) \simeq \cdots \simeq H^{\cdot}(M^{0},F) = 0.$$
 (4.20)

The first equality of (4.13) is a trivial consequence of (4.10). Let us show the second equality of (4.13).

We claim that

$$T_F(M) = \prod_{\gamma \in B} \left( T_{o(E_{\gamma}^u) \otimes_{\mathbf{R}} F|_{\gamma}}(\gamma) \right)^{(-1)^{\operatorname{ind}(\gamma)}}.$$
 (4.21)

Indeed, if F is unitary, using (4.16), (4.19), the Cheeger–Müller theorem [C79, M78], we can deduce (4.21) from the excision property of the Reidemeister torsion. If F is only unimodular, using Bismut–Zhang's theorem [BZ92, Theorem 0.2] and a property of Milnor torsion, (4.21) still holds true (see [ShY18, Proposition 2.13]).

By 
$$(1.13)$$
,  $(1.14)$ , and  $(4.21)$ , we get the second equality of  $(4.13)$ .

It is natural to ask if a similar result holds when the Morse–Smale flow has both fixed points and closed orbits. However, Sánchez-Morgado [SM96b] has shown that the heteroclinic orbits have a non-trivial contribution in the torsion invariant, and in this way he constructed a counterexample to the Fried conjecture on Seifert manifolds.

In [ShY18], Shen and Yu constructed a Milnor metric, which indeed contains the heteroclinic contributions, and obtained a proof for a modified version of the Fried conjecture for generally Morse–Smale flow with or without fixed points and for arbitrary flat vector bundle with arbitrary Hermitian metric. The formulation and the proof are based on [BZ92], using the determinant line of the cohomology. We refer the reader to [ShY18] for more details.

Let us mention that there is another interpretation of the Ruelle dynamical zeta function provided by Dang–Rivière [DaRi20c]. See also [DaRi19, DaRi20a, DaRi20b, DaRi21] for related works.

### 5. Geodesic flow

In this section, we discuss the Fried conjecture in the most interesting case the geodesic flow on the unit tangent bundle of a non-positively curved Riemannian manifold of dimension  $\geq 3$ . (The case of surface will be treated in Section 6.4.) In this case, the Fried conjecture can be proved formally via Bismut–Goette's V-invariant [BG04]. However, this argument remains non-rigorous due to the involved infinite-dimensional loop space. For locally symmetric spaces, we give the rigorous argument [F86a, MoSt91, Sh18] which is based on the Selberg trace formula and the Bismut orbital integral formula [B11].

This section is organized as follows. In Section 5.1, we study the flat vector bundle defined on the unit tangent bundle.

In Section 5.2, we study the geodesic flow on the unit tangent bundle of a non-positively curved Riemannian manifold.

In Section 5.3, we sketch the formal proof of the Fried conjecture via Bismut–Goette's V-invariant.

In Section 5.4, we introduce the reductive group, symmetric space, and the locally symmetric space. We state the main Theorem 5.9 of this section.

Sections 5.5–5.8 are devoted to sketching the proof of Theorem 5.9.

In Section 5.5, we recall the Selberg trace formula and the Bismut orbital integral formula, which are our main analytic tools.

In Section 5.6, we give a new proof of Moscovici–Stanton's vanishing theorem [MoSt91], which is due to Bismut [B11].

In Section 5.7, we sketch the proof of the Fried conjecture for the real reductive group of **R**-rank one, which is originally due to Fried [F86a].

In Section 5.8, we sketch the proof of the Fried conjecture for general reductive groups [Sh18].

### 5.1. Flat vector bundle on the unit tangent bundle

Let  $(Z, g^{TZ})$  be a compact Riemannian manifold of dimension m. Let

$$M = SZ = \{(x, v) \in TZ : |v| = 1\}$$
(5.1)

be the unit tangent bundle. Then  $\pi: M \to Z$  is a fibration of sphere  $\mathbb{S}^{m-1}$ .

**Proposition 5.1.** Assume dim  $Z \ge 3$ . Then all the flat vector bundles on M are the pull back of a flat vector bundle F on Z. Moreover, F is unitary (resp. unimodular) if and only if  $\pi^*F$  is unitary (resp. unimodular).

*Proof.* Consider the long exact sequence of homotopy groups

$$\cdots \to \pi_1(\mathbb{S}^{m-1}) \to \pi_1(M) \to \pi_1(Z) \to \pi_0(\mathbb{S}^{m-1}) \to \cdots$$
 (5.2)

associated to the fibration  $M \to Z$ . Since  $m \ge 3$ ,  $\pi_1(\mathbb{S}^{m-1}) = \pi_0(\mathbb{S}^{m-1}) = 1$ . By (5.2), we have the isomorphism of the groups

$$\pi_1(M) \simeq \pi_1(Z),\tag{5.3}$$

from which we deduce our proposition.

**Proposition 5.2.** Assume that Z is orientable and that dim  $Z \geqslant 3$ . Then  $\pi^*F$  is acyclic on M if and only if F is acyclic on Z. In additional, if dim Z is odd, then<sup>6</sup>

$$T_{\pi^*F}(M) = T_F(Z)^2.$$
 (5.4)

Proof. Let

$$\cdots \to H^{p-m}(Z,F) \to H^p(Z,F) \to H^p(M,\pi^*F) \to H^{p-m+1}(Z,F)$$
$$\to H^{p+1}(Z,F) \to \cdots$$
(5.5)

 $<sup>^6</sup>$ Since both of M and Z have odd dimensions, by Theorem 1.6, we do not need to specify the metric data to define the analytic torsion.

be the Gysin exact sequence [BoTu82, Proposition 14.33] associated to the fibration  $M \to Z$  of orientable (m-1)-spheres. Comparing the degrees in (5.5), for  $0 \le p \le m-2$  and  $m+1 \le q \le 2m-1$ , we have

$$H^p(M, \pi^*F) = H^p(Z, F), \qquad H^q(M, \pi^*F) = H^{q-m+1}(Z, F).$$
 (5.6)

By (5.6), using  $m \ge 3$ , we get the first statement of our proposition. By (1.16), we get (5.4).

**Remark 5.3.** Assume that dim  $Z \geqslant 3$  is odd. If we do not assume that Z is orientable, a detailed analysis on the Leray spectral sequence tells us that  $\pi^*F$  is acyclic on M if and only if F and  $o(TZ) \otimes_{\mathbf{R}} F$  are acyclic on Z. In this case, we have

$$T_{\pi^*F}(M) = T_F(Z) \cdot T_{o(TZ) \otimes_{\mathbf{R}} F}(Z). \tag{5.7}$$

Let us consider the geodesic flow on a negatively curved orientable surface  $(Z, g^{TZ})$  of genus  $g \ge 2$ , which will be used in Section 6.4. The long exact sequence of the homotopy groups (5.2) is simply

$$1 \to \mathbf{Z} \to \pi_1(M) \to \pi_1(Z) \to 1. \tag{5.8}$$

Let  $a_0 \in \pi_1(M)$  be a generator of  $\mathbf{Z} \subset \pi_1(M)$ . Let F be a flat vector bundle of rank r on M with holonomy  $\rho$ . Then  $\rho(a_0) \in \mathrm{GL}_r(\mathbf{C})$ .

**Corollary 5.4.** Let us assume that Z is a negatively curved orientable surface. Then  $H^{\cdot}(M, F) = 0$  if and only if 1 is not an eigenvalue of  $\rho(a_0)$ . In this case, we have

$$T_F(M) = \left| \det \left( \rho(a_0) \right) \right|^{(1-g)} \cdot \left| \det \left( 1 - \rho(a_0) \right) \right|^{2(g-1)}.$$
 (5.9)

*Proof.* We use the Leray spectral sequence associated to the fibration  $M \to Z$  as in the proof of Propositions 3.2. By [BoTu82, Theorem 14.18],  $(E_r^{p,q}, d_r)_{r\geqslant 0}$  converges to  $H^{\cdot}(M, F)$ , and

$$E_2^{p,q} = H^p(Z, H^q(Y, F|_Y)). (5.10)$$

Since the fibre Y is a circle, for  $r \ge 2$  and  $q \ne 0$  or 1, we see that  $E_r^{p,q} = 0$ . Since  $d_r$  sends  $E_r^{p,q}$  to  $E_r^{p+r,q-r+1}$ . We get  $d_3 = d_4 = \cdots = 0$ , and  $d_2$  vanishes except

$$d_2: E_2^{0,1} \to E_2^{2,0}. (5.11)$$

Indeed, by [BoTu82, p. 178], up to a sign,  $d_2|_{E_2^{0,1}}$  is the multiplication by the Euler characteristic class of Z, which is an isomorphism since  $g \ge 2$ . So the spectral sequence degenerates at r = 3 such that

$$H^{0}(M,F) = H^{0}(Z,H^{0}(Y,F|_{Y})), \quad H^{1}(M,F) = H^{1}(Z,H^{0}(Y,F|_{Y})),$$
  

$$H^{2}(M,F) = H^{0}(Z,H^{1}(Y,F|_{Y})), \quad H^{3}(M,F) = H^{2}(Z,H^{1}(Y,F|_{Y})).$$
(5.12)

Since Y is a circle and since Z is orientable, we have an isomorphism of vector bundles on Z,

$$H^1(Y, F|_Y) \simeq H^0(Y, F|_Y).$$
 (5.13)

By (5.12) and (5.13), we see that F is acyclic on M if and only if  $H^0(Y, F|_Y)$  is acyclic on Z. By the Gauss–Bonnet theorem,

$$\chi(Z, H^0(Y, F|_Y)) = 2\operatorname{rk}[H^0(Y, F|_Y)](1 - g), \tag{5.14}$$

where  $\operatorname{rk}[H^0(Y, F|_Y)]$  denotes the rank of the vector bundle  $H^0(Y, F|_Y)$ . By (5.14), we see that  $H^0(Y, F|_Y)$  is acyclic on Z if and only if  $H^0(Y, F|_Y) = 0$ . The latter means exactly that 1 is not an eigenvalue of  $\rho(a_0)$ . By (1.13), (1.14), and (1.15), we get (5.9).

**Remark 5.5.** Equation (5.9) is obtained in [F86b] by a purely topological method under the assumption that F is unitarily flat.

### 5.2. Geodesic flow of the non-positively curved manifolds

Let  $(Z, g^{TZ})$  be a Riemannian manifold with non-positive sectional curvature. Let  $(\phi_t)_{t \in \mathbf{R}}$  be the geodesic flow on the unit tangent bundle M = SZ. Let  $V \in C^{\infty}(M, TM)$  be the generator of  $\phi$ .

Let  $LZ = C^{\infty}(\mathbb{S}^1, Z)$  be the free loop space of Z. It is equipped with a canonical  $\mathbb{S}^1$ -action given by

$$s \cdot x = x_{+s},$$
 for  $s \in \mathbb{S}^1, x \in LZ.$  (5.15)

Denote by  $\Gamma$  the fundamental group of Z. Let  $[\Gamma]$  be the set of conjugacy classes of  $\Gamma$ . We identify  $[\Gamma]$  with the set of freely homotopy classes of loops on Z. For  $[\gamma] \in [\Gamma]$ , let  $(LZ)_{[\gamma]} \subset LZ$  be the subset of LZ formed by all the loops on Z with the freely homotopy class  $[\gamma]$ . Write

$$LZ = \coprod_{[\gamma] \in [\Gamma]} (LZ)_{[\gamma]}. \tag{5.16}$$

Let E be the  $\mathbb{S}^1$ -invariant energy functional on LZ defined by

$$E: x. \in LZ \to \frac{1}{2} \int_0^1 |\dot{x}_s|^2 ds.$$
 (5.17)

Then E is a convex function whose critical points are closed geodesics. Let  $B_{[\gamma]}$  be the critical points set of E in  $(LZ)_{[\gamma]}$ . Since  $(Z, g^{TZ})$  has non-positive sectional curvature,

$$B_{[1]} \simeq Z. \tag{5.18}$$

Let us give a more explicit description of  $B_{[\gamma]}$ . Take an element  $\gamma$  in  $[\gamma]$ . Up to evident isomorphisms, our objects constructed below do not depend on the choice of  $\gamma$  in  $[\gamma]$ . Let X be the universal cover of Z. Let  $g^{TX}$  be the induced Riemannian metric on X by  $g^{TZ}$ . Let  $d_X(\cdot,\cdot)$  be the Riemannian distance on X. Since Z and also X have non-positive sectional curvature, the displacement function  $x \in X \to d_X(x,\gamma x)$  is convex. Let  $\ell_{[\gamma]} \in \mathbf{R}_+$  be the minimal value  $^7$  of the displacement function on X and let  $X(\gamma) \subset X$  be its minimal set. Since Z is compact and since the displacement function is convex,  $X(\gamma)$  is a non-empty

<sup>&</sup>lt;sup>7</sup>As the notation indicates,  $\ell_{[\gamma]}$  does not depend on the choice of  $\gamma \in [\gamma]$ .

convex set (see [BaGrSch85, Section 6], [Ma17, Proposition 3.9]). Let  $\Gamma(\gamma)$  be the centralizer of  $\gamma$  in  $\Gamma$ . Then  $\Gamma(\gamma)$  acts on  $X(\gamma)$ . We have the identification

$$x. \in B_{[\gamma]} \simeq [\widetilde{x}_0] \in \Gamma(\gamma) \backslash X(\gamma),$$
 (5.19)

where  $\widetilde{x}$  is the lifting of x in X. By the convexity of  $X(\gamma)$ , we see that  $B_{[\gamma]}$  is connected. Recall that M = SZ. We can also identify  $B_{[\gamma]}$  with a connected component of  $M^{\phi_{\ell_{[\gamma]}}}$  by

$$x. \in B_{[\gamma]} \to (x_0, \dot{x}_0/\ell_{[\gamma]}) \in M^{\phi_{\ell_{[\gamma]}}}. \tag{5.20}$$

Since  $M^{\phi_{\ell_{[\gamma]}}}$  is compact, we see that  $B_{[\gamma]}$  is compact. By (5.20), equations (2.14) and (2.15) become

$$\wp(\phi_{\cdot}) = \coprod_{[\gamma] \in [\Gamma] - \{1\}} B_{[\gamma]} \times \{\ell_{[\gamma]}\}, \quad \overline{\wp}(\phi_{\cdot}) = \coprod_{[\gamma] \in [\Gamma] - \{1\}} B_{[\gamma]} / \mathbb{S}^1 \times \{\ell_{[\gamma]}\}. \quad (5.21)$$

**Proposition 5.6.** Assume that  $(Z, g^{TZ})$  has non-positive sectional curvature. The following statements hold.

- (1) Condition (1) of Assumption 2.4 holds.
- (2) Condition (2) of Assumption 2.4 holds if Z is a negatively curved manifold or if Z is a locally symmetric space.

*Proof.* For Condition (1), it is enough to show that for  $r \ge 0$  the set

$$\{ [\gamma] \in [\Gamma] : \ell_{[\gamma]} \leqslant r \} \tag{5.22}$$

is finite. Fix a point  $x \in X$  in the universal cover X. Fix a fundamental domain  $F_Z \subset X$  of Z in X. There is  $z_0 \in F_Z$  and  $\gamma_1 \in \Gamma$  such that

$$\ell_{[\gamma]} = d_X(\gamma_1 z_0, \gamma \gamma_1 z_0). \tag{5.23}$$

By (5.23) and by the triangular inequality, we have

$$d_X(\gamma_1 x, \gamma \gamma_1 x) \leqslant \ell_{[\gamma]} + 2 \max_{z \in F_Z} d(x, z). \tag{5.24}$$

Using (5.24), we get

$$|\{[\gamma] \in [\Gamma] : \ell_{[\gamma]} \leqslant r\}| \leqslant |\{\gamma \in \Gamma : d_X(x, \gamma x) \leqslant r + 2 \max_{z \in F_Z} d(x, z)\}|.$$
 (5.25)

It is known by [Mi68b, Remark p.1, Lemma 2] that the set on the right-hand side of (5.25) is finite<sup>8</sup>.

The statement (2) is clear if Z is a negatively curved manifold. When Z is a locally symmetric space, the statement (2) is a consequence of [B11, (3.3.17), (3.5.10)].

### 5.3. The V-invariant and a formal proof of the Fried conjecture

Let us give a formal proof of (2.22) using the V-invariant of Bismut-Goette [BG04].

<sup>&</sup>lt;sup>8</sup>Indeed, it is smaller than  $Ce^{Cr}$  for certain C > 0.

**5.3.1. The** *V*-invariant. Let *S* be a closed manifold equipped with an action of a compact Lie group *L*, with Lie algebra  $\mathfrak{l}$ . If  $a \in \mathfrak{l}$ , let  $a^S$  be the corresponding vector field on *S*. Bismut–Goette [BG04] introduced the *V*-invariant  $V_a(S) \in \mathbf{R}$ .

Let f be an  $a^S$ -invariant Morse–Bott function on S. Let  $B_f \subset S$  be the critical submanifold. Since  $a^S|_{B_f} \in TB_f$ ,  $V_a(B_f)$  is also well defined. By [BG04, Theorem 4.10],  $V_a(S)$  and  $V_a(B_f)$  are related by a simple formula.

**5.3.2.** A formal proof of the Fried conjecture. Let us argue formally as in [Sh18, Section 1E]. Let a be the generator of the Lie algebra of  $\mathbb{S}^1$  such that  $\exp(a) = 1$ .

As explained in [B05, Equation (0.3)], if F is a unitarily flat vector bundle on Z such that  $H^{\cdot}(Z, F) = 0$ , at least formally, we have

$$\log T_F(Z) = -\sum_{[\gamma] \in [\Gamma]} \operatorname{Tr}[\rho(\gamma)] V_a((LZ)_{[\gamma]}). \tag{5.26}$$

Let  $(Z, g^{TZ})$  be an odd-dimensional oriented Riemannian manifold with non-positive sectional curvature. Assume that the energy functional is Morse–Bott<sup>9</sup>. Applying [BG04, Theorem 4.10] to the infinite-dimensional manifold  $(LZ)_{[\gamma]}$  with the  $\mathbb{S}^1$ -invariant Morse–Bott functional E, we have the formal identity

$$V_a((LZ)_{[\gamma]}) = V_a(B_{[\gamma]}). \tag{5.27}$$

Since  $B_{[1]} \simeq Z$  is formed by the trivial geodesics, by the definition of the V-invariant,

$$V_a(B_{[1]}) = 0. (5.28)$$

By [BG04, Proposition 4.26], if  $[\gamma] \in [\Gamma] - \{1\}$ , then

$$V_a(B_{[\gamma]}) = -\frac{\chi_{\text{orb}}(B_{[\gamma]}/\mathbb{S}^1)}{2m_{[\gamma]}}.$$
(5.29)

By Proposition 5.2 and by (5.26)–(5.29), we get a formal identity

$$\log T_{\pi^* F}(M) = 2\log T_F(Z) = \sum_{[\gamma] \in [\Gamma] - \{1\}} \text{Tr}[\rho(\gamma)] \frac{\chi_{\text{orb}}(B_{[\gamma]}/\mathbb{S}^1)}{m_{[\gamma]}}, \tag{5.30}$$

which is formally just  $(2.22)^{10}$ .

### 5.4. Reductive group, globally and locally symmetric space

We give a rigorous argument in the case of locally symmetric space of reductive type. Let us recall some preliminaries that are needed.

 $<sup>^9</sup>$ Since this part is completely formal, we will not give the precise definition of the Morse–Bott function on the loop space LZ. Here we can understand it by the fact that the energy functional is convex and its critical points form a smooth manifold.

<sup>&</sup>lt;sup>10</sup>We need also to show that the sign that appeared in the Fuller index is positive when  $\chi_{\rm orb}(B_{[\gamma]}/\mathbb{S}^1) \neq 0$ . Indeed, it follows from the fact that Z has an odd dimension and is orientable, and the fact that  $\chi_{\rm orb}(B_{[\gamma]}/\mathbb{S}^1) = 0$  if dim  $B_{[\gamma]}$  is even. We omit the details. For negatively curved manifolds, a proof can be found in [GiLiPo13, Appendix B].

**5.4.1. Reductive group.** Let G be a linear connected real reductive group [K86, p. 3], and let  $\theta \in \text{Aut}(G)$  be the Cartan involution. This means that G is a closed connected group of real matrices that is stable under transpose, and  $\theta$  is the composition of transpose and inverse of matrices. Let K be the maximal compact subgroup of G which is the fixed point set of  $\theta$ .

Let  $\mathfrak{g}$  be the Lie algebra of G, and let  $\mathfrak{k} \subset \mathfrak{g}$  be the Lie algebra of K. The Cartan involution  $\theta$  acts naturally as a Lie algebra automorphism of  $\mathfrak{g}$ . Then  $\mathfrak{k}$  is the eigenspace of  $\theta$  associated with the eigenvalue 1. Let  $\mathfrak{p}$  be the eigenspace with the eigenvalue -1, so that

$$\mathfrak{g} = \mathfrak{p} \oplus \mathfrak{k}. \tag{5.31}$$

We have

$$[\mathfrak{k},\mathfrak{k}] \subset \mathfrak{k}, \qquad [\mathfrak{p},\mathfrak{p}] \subset \mathfrak{k}, \qquad [\mathfrak{k},\mathfrak{p}] \subset \mathfrak{p}.$$
 (5.32)

By [K86, Proposition 1.2], we have the diffeomorphism

$$(Y,k) \in \mathfrak{p} \times K \to e^Y k \in G.$$
 (5.33)

Let B be a real-valued non-degenerate bilinear symmetric form on  $\mathfrak g$  which is invariant under the adjoint action Ad of G on  $\mathfrak g$ , and also under  $\theta$ . Then (5.31) is an orthogonal splitting of  $\mathfrak g$  with respect to B. We assume B to be positive on  $\mathfrak p$ , and negative on  $\mathfrak k$ . The form  $\langle \cdot, , \cdot \rangle = -B(\cdot, \theta \cdot)$  defines an  $\mathrm{Ad}(K)$ -invariant scalar product on  $\mathfrak g$  such that the splitting (5.31) is still orthogonal. We denote by  $|\cdot|$  the corresponding norm.

The real (resp. complex) rank of G, denoted by  $\operatorname{rk}_{\mathbf{R}} G$  (resp.  $\operatorname{rk}_{\mathbf{C}} G$ ), is defined by the dimension of the maximal abelian subspace of  $\mathfrak{p}$  (resp. the Cartan subalgebra of  $\mathfrak{g}$ ). The fundamental rank of G is defined by

$$\delta(G) = \operatorname{rk}_{\mathbf{C}} G - \operatorname{rk}_{\mathbf{C}} K \in \mathbf{N}. \tag{5.34}$$

Clearly,  $\delta(\mathfrak{g}) \in \mathbf{N}$  is also well defined. The following proposition is a consequence of the classification theory of real simple Lie algebras (see [B11, Remark 7.9.2]).

**Theorem 5.7.** The only simple Lie algebras<sup>11</sup>  $\mathfrak{g}$  with  $\delta(\mathfrak{g}) = 1$  are given by

$$\mathfrak{sl}_3(\mathbf{R}) \ or \ \mathfrak{so}(p,q) \ with \ pq > 1 \ odd.$$
 (5.35)

In Table 1, we list some reductive groups G, the maximal compact subgroups K, the dimensions  $\mathfrak{p}$ , and the fundamental ranks  $\delta(G)$ .

**5.4.2. Semisimple elements.** If  $\gamma \in G$ , we denote by  $Z(\gamma) \subset G$  the centralizer of  $\gamma$  in G, and by  $\mathfrak{z}(\gamma) \subset \mathfrak{g}$  its Lie algebra. If  $a \in \mathfrak{g}$ , let  $Z(a) \subset G$  be the stabilizer of a in G, and let  $\mathfrak{z}(a) \subset \mathfrak{g}$  be its Lie algebra.

An element  $\gamma \in G$  is said to be semisimple if  $\gamma$  can be conjugate to  $e^a k^{-1}$  such that

$$a \in \mathfrak{p}, \qquad k \in K, \qquad \mathrm{Ad}(k)a = a. \tag{5.36}$$

<sup>&</sup>lt;sup>11</sup>We have also  $\delta(\mathbf{R}) = 1$ . But the abelian Lie algebra  $\mathbf{R}$  is not considered as simple Lie algebra.

| G                   | R   | $\mathrm{GL}_n^+(\mathbf{R})$  | $\mathrm{SL}_n(\mathbf{R})$    | $SO^0(p,q)$  |
|---------------------|-----|--------------------------------|--------------------------------|--|
| K                   | {0} | SO(n)                          | SO(n)                          | $SO(p) \times SO(q)$   |
| $\dim \mathfrak{p}$ | 1   | $\frac{n(1+n)}{2}$             | $\frac{n(1+n)}{2} - 1$         | pq   |
| $\delta(G)$         | 1   | $n - \left[\frac{n}{2}\right]$ | $n-1-\left[\frac{n}{2}\right]$ | $\left[\frac{p+q}{2}\right] - \left[\frac{p}{2}\right] - \left[\frac{q}{2}\right]$ |

Table 1. Examples of reductive Lie group.

Let  $\gamma = e^a k^{-1}$  be such that (5.36) holds. By [B11, (3.3.4), (3.3.6)], we have

$$Z(\gamma) = Z(a) \cap Z(k),$$
  $\mathfrak{z}(\gamma) = \mathfrak{z}(a) \cap \mathfrak{z}(k).$  (5.37)

Set

$$\mathfrak{p}(\gamma) = \mathfrak{z}(\gamma) \cap \mathfrak{p}, \qquad \mathfrak{k}(\gamma) = \mathfrak{z}(\gamma) \cap \mathfrak{k}, \qquad K(\gamma) = Z(\gamma) \cap K. \tag{5.38}$$

From (5.37) and (5.38), we get

$$\mathfrak{z}(\gamma) = \mathfrak{p}(\gamma) \oplus \mathfrak{k}(\gamma). \tag{5.39}$$

By [K02, Proposition 7.25],  $Z(\gamma)$  is a reductive subgroup (not necessarily connected) of G with maximal compact subgroup  $K(\gamma)$ , and with Cartan decomposition (5.39).

**5.4.3. The symmetric space.** Set X = G/K to be the associated symmetric space. Then

$$p: G \to X = G/K \tag{5.40}$$

is a K-principle bundle.

Let  $\tau$  be a finite-dimensional orthogonal representation of K on the real Euclidean space  $E_{\tau}$ . Then  $\mathcal{E}_{\tau} = G \times_K E_{\tau}$  is a real Euclidean vector bundle on X. The space of smooth sections  $C^{\infty}(X, \mathcal{E}_{\tau})$  on X can be identified with the subspace  $C^{\infty}(G, E_{\tau})^K$  of K-invariant smooth  $E_{\tau}$ -valued functions on G.

By adjoint action, K acts isometrically on  $\mathfrak{p}.$  Using the above construction, the tangent bundle of X is given by

$$TX = G \times_K \mathfrak{p}. \tag{5.41}$$

The bilinear form  $B|_{\mathfrak{p}}$  induces a G-invariant Riemannian metric  $g^{TX}$  on X. It is well known that  $(X,g^{TX})$  is a Riemannian manifold of non-positive sectional curvature. For  $x,y\in X$ , we denote by  $d_X(x,y)$  the Riemannian distance on X.

**5.4.4. The locally symmetric space.** Let  $\Gamma \subset G$  be a discrete torsion-free cocompact subgroup of G. Take  $Z = \Gamma \backslash X = \Gamma \backslash G/K$ . Then Z is a connected closed orientable Riemannian locally symmetric manifold with non-positive sectional curvature. Since X is contractible,  $\pi_1(Z) = \Gamma$  and X is the universal cover of Z.

By [Sel60, Lemmas 1],  $\Gamma$  contains the identity element and non-elliptic semisimple elements. Recall that  $\Gamma(\gamma)$  is the centralizer of  $\gamma$  in  $\Gamma$  defined before (5.19). By [Sel60, Lemma 2],  $\Gamma(\gamma)$  is cocompact in  $Z(\gamma)$ . The following proposition is [DuKV79, Proposition 5.15], relating the geometric object considered in Section 5.2 and the group theoretic object considered in this section.

**Proposition 5.8.** For  $[\gamma] \in [\Gamma] - \{1\}$ , we have the following identifications,

$$\ell_{[\gamma]} = |a|, \qquad X(\gamma) \simeq Z(\gamma)/K(\gamma), \qquad B_{[\gamma]} \simeq \Gamma(\gamma) \backslash Z(\gamma)/K(\gamma).$$
 (5.42)

Let us state the main theorem of [Sh18, Theorem 1.1], which generalizes [F86a, Theorem 3] in the case where  ${\rm rk}_{\bf R}[G]=1$  and [MoSt91, Corollary 2.2] in the case where  $\delta(G)\neq 1$ .

**Theorem 5.9.** Assume that  $Z = \Gamma \backslash G/K$  and dim  $Z \geqslant 3$  is odd. Let F be a unitarily flat vector bundle on Z. The following statements hold.

- (1) The dynamical zeta function  $R_{\phi,\rho}(\sigma)$  is well defined in the sense of Definition 2.7
- (2) There exist explicit constants  $C_{\rho} \in \mathbf{R}^*$  and  $r_{\rho} \in \mathbf{Z}$  (see (5.99)) such that, when  $\sigma \to 0$ ,

$$R_{\phi,\rho}(\sigma) = C_{\rho} T_F(Z)^2 \sigma^{r_{\rho}} + \mathcal{O}\left(\sigma^{r_{\rho}+1}\right). \tag{5.43}$$

(3) If  $H^{\cdot}(Z, F) = 0$ , then

$$C_{\rho} = 1,$$
  $r_{\rho} = 0,$  (5.44)

so that

$$R_{\phi,\rho}(0) = T_F(Z)^2 = T_{\pi^*(F)}(M).$$
 (5.45)

We sketch the proof of Theorem 5.9 in Sections 5.5–5.8.

**Remark 5.10.** If we do not require  $\Gamma$  to be torsion free, then  $Z = \Gamma \backslash G/K$  is an orbifold. In [ShY17], we show that the above theorem still holds true.

**Remark 5.11.** Using Hirzebruch proportionality [H66, Theorem 22.3.1] and Bott's formula [Bo65, p. 175], we can show that if  $\delta(G) \ge 2$ , then for all  $[\gamma] \in [\Gamma] - \{1\}$ ,

$$\chi(B_{[\gamma]}/\mathbb{S}^1) = 0. \tag{5.46}$$

So  $R_{\phi,\rho}(\sigma) \equiv 1$ .

Remark 5.12. We do not need to study the case where  $Z = \Gamma \backslash G/K$  with dim  $Z \geqslant 4$  is even. Recall that  $\delta(G)$  and dim Z have the same parity. If  $\delta(G) \geqslant 2$ , by (1.10) and Remark 5.11, we have  $R_{\phi,\rho}(0) = T_F(Z) = 1$ . If  $\delta(G) = 0$ , there are no acyclic flat vector bundles on Z. Indeed, as in (5.46), by Hirzebruch proportionality [H66, Theorem 22.3.1] and Bott's formula [Bo65, p. 175], we can deduce that

$$(-1)^{\frac{1}{2}\dim Z}\chi(Z) > 0. \tag{5.47}$$

Using the Gauss-Bonnet-Chern Theorem, by (5.47), we see that

$$\chi(Z, F) = \operatorname{rk}[F]\chi(Z) \neq 0, \tag{5.48}$$

which implies  $H^{\cdot}(Z, F) \neq 0$ .

Remark 5.13. By Remark 5.11, if dim Z is odd and  $\delta(G) \ge 3$ , to show the Fried conjecture, it is enough to show  $T_F(Z) = 1$ , which is known as the Moscovici–Stanton vanishing theorem (see Section 5.6).

### 5.5. The Selberg trace formula

In this section, we recall the Selberg trace formula. We introduce the Casimir operator and its heat kernel, orbital integral, and an explicit orbital integral formula [B11, Theorem 6.1.1] for the heat operator of the Casimir.

**5.5.1. Casimir operator.** Let  $\mathscr{U}(\mathfrak{g})$  be the enveloping algebra of  $\mathfrak{g}$ . We identify  $\mathscr{U}(\mathfrak{g})$  with the algebra of left-invariant differential operators on G. Let  $C^{\mathfrak{g}} \in \mathscr{U}(\mathfrak{g})$  be the Casimir element of  $(\mathfrak{g}, B)$ . If  $\{e_1, \ldots, e_{\dim \mathfrak{p}}\}$  is an orthonormal basis of  $\mathfrak{p}$ , and if  $\{f_1, \ldots, f_{\dim \mathfrak{k}}\}$  is an orthonormal basis of  $\mathfrak{k}$ , then

$$C^{\mathfrak{g}} = -\sum_{i=1}^{\dim \mathfrak{p}} e_i^2 + \sum_{i=1}^{\dim \mathfrak{k}} f_i^2.$$
 (5.49)

It is well known that  $C^{\mathfrak{g}}$  is in the center of  $\mathscr{U}(\mathfrak{g})$ .

We define  $C^{\mathfrak{k}}$  in the same way. Let  $\tau$  be a finite-dimensional representation of K on  $E_{\tau}$ . We denote by  $C^{\mathfrak{k},E_{\tau}} \in \operatorname{End}(E_{\tau})$  the corresponding Casimir operator acting on  $E_{\tau}$ , so that

$$C^{\ell, E_{\tau}} = \sum_{i=1}^{\dim \ell} \tau(f_i)^2. \tag{5.50}$$

Let  $C^{\mathfrak{g},X,\tau}$  be the Casimir element of G acting on  $C^{\infty}(X,\mathcal{E}_{\tau})$ . Then  $C^{\mathfrak{g},X,\tau}$  is a formally self-adjoint second-order elliptic differential operator which is bounded from below. If  $E_{\tau} = \Lambda^{\cdot}(\mathfrak{p}^{*})$ , then  $C^{\infty}(X,\mathcal{E}_{\tau}) = \Omega^{\cdot}(X)$ . In this case, we write  $C^{\mathfrak{g},X} = C^{\mathfrak{g},X,\tau}$ . By [B11, Proposition 7.8.1],  $C^{\mathfrak{g},X}$  coincides with the Hodge Laplacian acting on  $\Omega^{\cdot}(X)$ .

We denote by  $\widehat{p}: \Gamma \backslash G \to Z$  and  $\pi: X \to Z$  the natural projections, so that the diagram

$$G \longrightarrow \Gamma \backslash G$$

$$\downarrow^{p} \qquad \downarrow^{\widehat{p}}$$

$$X \xrightarrow{\pi} Z$$

$$(5.51)$$

commutes. Recall that the group  $\Gamma$  acts isometrically on the left on X. This action lifts to all the homogeneous Euclidean vector bundles  $\mathcal{E}_{\tau}$  constructed in Subsection 5.4.3. It descends to a Euclidean vector bundle  $\mathcal{F}_{\tau} = \Gamma \backslash \mathcal{E}_{\tau}$  on Z. Let F be the unitarily flat vector bundle of rank r on Z. Let  $\rho : \Gamma \to U(r)$  be the holonomy of F. Let  $C^{\mathfrak{g},Z,\tau,\rho}$  be the Casimir element of G acting on  $C^{\infty}(Z,\mathcal{F}_{\tau}\otimes_{\mathbf{R}}F)$ . As before, when  $E_{\tau} = \Lambda^{\cdot}(\mathfrak{p}^{*})$ , we write  $C^{\mathfrak{g},Z,\rho} = C^{\mathfrak{g},Z,\tau,\rho}$ . Then,

$$\Box^Z = C^{\mathfrak{g}, Z, \rho}. \tag{5.52}$$

**5.5.2. The heat kernel of**  $C^{\mathfrak{g},X,\tau}$ **.** We fix a finite-dimensional real orthogonal representation  $(\tau, E_{\tau})$  of K.

Let  $p_t^{X,\tau}(x,x')$  be the smooth kernel of  $\exp(-tC^{\mathfrak{g},X,\tau}/2)$  with respect to the Riemannian volume  $dv_X$  on X. The Gaussian estimate on the heat kernel tells us that for t>0, there exist c>0 and C>0 such that for  $x,x'\in X$ ,

$$\left| p_t^{X,\tau}(x,x') \right| \leqslant C \exp\left( -c \ d_X^2(x,x') \right). \tag{5.53}$$

Set

$$p_t^{X,\tau}(g) = p_t^{X,\tau}(p_1, p_g).$$
 (5.54)

It is a smooth  $\operatorname{End}(E_{\tau})$ -valued function on G. For  $g \in G$  and  $k, k' \in K$ , we have

$$p_t^{X,\tau}(kgk') = \tau(k)p_t^{X,\tau}(g)\tau(k'). \tag{5.55}$$

We can recover  $p_t^{X,\tau}(x,x')$  by

$$p_t^{X,\tau}(x,x') = p_t^{X,\tau}(g^{-1}g'), \tag{5.56}$$

where  $g, g' \in G$  are such that pg = x, pg' = x'.

In the sequel, we do not distinguish  $p_t^{X,\tau}(x,x')$  and  $p_t^{X,\tau}(g)$ . We refer to both of them as the smooth kernel of  $\exp(-tC^{\mathfrak{g},X,\tau}/2)$ .

**5.5.3.** The Selberg trace formula. We will write the trace of  $\exp(-tC^{\mathfrak{g},Z,\tau,\rho}/2)$  with the help of the heat kernel  $p_t^{X,\tau}(g)$  on X. Let  $dv_K$  be the Haar measure on K such that the volume of K is equal to 1. By (5.33),  $dv_K$  and  $B|_{\mathfrak{p}}$  induce a Haar measure  $dv_G$  on G. Let  $dv_{Z(\gamma)}$  be the Haar measure on  $Z(\gamma)$  constructed in the same way. Let  $dv_{Z(\gamma)\setminus G}$  be the right invariant measure on  $Z(\gamma)\setminus G$  such that

$$dv_G = dv_{Z(\gamma)} dv_{Z(\gamma)\backslash G}. (5.57)$$

It is known (see [B11, (3.4.36)]) that for  $s \gg 1$ ,

$$\int_{g \in Z(\gamma) \backslash G} e^{-sd_X(p\gamma g, pg)} dv_{Z(\gamma) \backslash G} < \infty.$$
 (5.58)

**Definition 5.14.** Let  $\gamma \in G$  be semisimple. We define the orbital integral of  $\exp(-tC^{\mathfrak{g},X,\tau}/2)$  by

$$\operatorname{Tr}^{[\gamma]}\left[\exp\left(-tC^{\mathfrak{g},X,\tau}/2\right)\right] = \int_{g\in Z(\gamma)\backslash G} \operatorname{Tr}^{E_{\tau}}\left[p_t^{X,\tau}(g^{-1}\gamma g)\right] dv_{Z(\gamma)\backslash G}. \tag{5.59}$$

By (5.53) and (5.58), we see that the integration in (5.59) is well defined.

**Theorem 5.15.** There exist c > 0, C > 0 such that for t > 0, we have

$$\sum_{[\gamma]\in[\Gamma]-\{1\}} \operatorname{vol}\left(B_{[\gamma]}\right) \left| \operatorname{Tr}^{[\gamma]}\left[\exp\left(-tC^{\mathfrak{g},X,\tau}/2\right)\right] \right| \leqslant C \exp\left(-\frac{c}{t} + Ct\right). \tag{5.60}$$

For t > 0, the following identity holds,

$$\operatorname{Tr}\left[\exp\left(-tC^{\mathfrak{g},Z,\tau,\rho}/2\right)\right] = \sum_{[\gamma]\in[\Gamma]} \operatorname{vol}\left(B_{[\gamma]}\right) \operatorname{Tr}[\rho(\gamma)] \operatorname{Tr}^{[\gamma]}\left[\exp(-tC^{\mathfrak{g},X,\tau}/2)\right]. \tag{5.61}$$

*Proof.* The estimate in (5.60) and the convergence of (5.61) are consequences of the following fact: there exist c > 0, C > 0 such that for t > 0 and  $x \in X$ , we have

$$\sum_{\gamma \in \Gamma - \{1\}} \left| p_t^{X,\tau}(x,\gamma x) \right| \leqslant C \exp\left(-\frac{c}{t} + Ct\right). \tag{5.62}$$

The proof of (5.62) can be found in [Sh18, Proposition 4.8] (see also [MaMari15, Theorem 4]), where we use an estimate similar to the one in (5.25).

Let us prove (5.61). We disregard the convergence problem. Let  $p_t^{Z,\tau,\rho}(z,z')$  be the smooth kernel of  $\exp(-tC^{\mathfrak{g},Z,\tau,\rho}/2)$  with respect to the Riemannian volume on Z. For  $x \in X$  and  $g \in G$  such that pg = x, we have

$$p_t^{Z,\tau,\rho}\left(\pi x,\pi x\right) = \sum_{\gamma\in\Gamma} \rho(\gamma) p_t^{X,\tau}\left(g^{-1}\gamma g\right). \tag{5.63}$$

Recall that  $F_Z \subset X$  is the fundamental domain of Z in X. Then,  $p^{-1}(F_Z) \subset G$  is a fundamental domain of  $\Gamma \setminus G$  in G. Since vol(K) = 1, by (5.63), we have

$$\operatorname{Tr}\left[\exp\left(-tC^{\mathfrak{g},Z,\tau,\rho}/2\right)\right] = \int_{g\in p^{-1}(F_Z)} \sum_{\gamma\in\Gamma} \operatorname{Tr}\left[\rho(\gamma)\right] \operatorname{Tr}\left[p_t^{X,\tau}\left(g^{-1}\gamma g\right)\right] dv_G. \quad (5.64)$$

Take  $\gamma \in \Gamma$ . Recall that  $[\gamma] \in [\Gamma]$  is the conjugacy class of  $\gamma$  in  $\Gamma$ . We claim that

$$\int_{g \in p^{-1}(F_Z)} \sum_{\gamma' \in [\gamma]} \operatorname{Tr} \left[ p_t^{X,\tau} \left( g^{-1} \gamma g \right) \right] dv_G = \operatorname{vol}(\Gamma(\gamma) \backslash Z(\gamma)) \operatorname{Tr}^{[\gamma]} \left[ \exp(-t C^{\mathfrak{g},X,\tau}/2) \right].$$
(5.65)

Indeed, since  $\gamma' \to (\gamma')^{-1} \gamma \gamma'$  induces an identification of  $\Gamma(\gamma) \backslash \Gamma \simeq [\gamma]$ , we have

$$\sum_{\gamma' \in [\gamma]} \operatorname{Tr} \left[ p_t^{X,\tau} \left( g^{-1} \gamma' g \right) \right] = \sum_{\gamma' \in \Gamma(\gamma) \setminus \Gamma} \operatorname{Tr} \left[ p_t^{X,\tau} \left( g^{-1} (\gamma')^{-1} \gamma \gamma' g \right) \right]. \tag{5.66}$$

By (5.66), and by changing variable, we have

$$\int_{g \in p^{-1}(F_Z)} \sum_{\gamma' \in [\gamma]} \operatorname{Tr} \left[ p_t^{X,\tau} \left( g^{-1} \gamma' g \right) \right] dv_G \tag{5.67}$$

$$= \int_{\bigcup_{\gamma' \in \Gamma(\gamma) \backslash \Gamma} \gamma' p^{-1}(F_Z)} \operatorname{Tr} \left[ p_t^{X,\tau} \left( g^{-1} \gamma g \right) \right] dv_G = \int_{\Gamma(\gamma) \backslash G} \operatorname{Tr} \left[ p_t^{X,\tau} \left( g^{-1} \gamma g \right) \right] dv_{\Gamma(\gamma) \backslash G},$$

where in the last equality we use the fact that  $\bigcup_{\gamma' \in \Gamma(\gamma) \setminus \Gamma} \gamma' p^{-1}(F_Z)$  is a fundamental domain of  $\Gamma(\gamma) \setminus G$  in G. Using the fibration  $\Gamma(\gamma) \setminus G \to Z(\gamma) \setminus G$ , and using the fact that the integrand  $\operatorname{Tr}\left[p_t^{X,\tau}\left(g^{-1}\gamma g\right)\right]$  is constant on the fibre, by (5.59) and (5.67), we get (5.65).

By Proposition 5.8, (5.64), and (5.65), using  $vol(K(\gamma)) = 1$ , we get (5.61).  $\square$ 

**5.5.4. Bismut orbital integral formula.** Let us recall an explicit orbital integral formula for  $\operatorname{Tr}^{[\gamma]}\left[\exp(-tC^{\mathfrak{g},X,\tau}/2)\right]$  obtained by Bismut [B11, Theorem 6.1.1].

Let  $\gamma = e^a k^{-1} \in G$  be semisimple as in (5.36). Set

$$\mathfrak{p}_0 = \ker(\mathrm{ad}(a)) \cap \mathfrak{p}, \qquad \mathfrak{k}_0 = \ker(\mathrm{ad}(a)) \cap \mathfrak{k}, \qquad \mathfrak{z}_0 = \ker(\mathrm{ad}(a)) \cap \mathfrak{z}.$$
 (5.68)

Let  $\mathfrak{p}_0^{\perp} \subset \mathfrak{p}$ ,  $\mathfrak{k}_0^{\perp} \subset \mathfrak{k}$ ,  $\mathfrak{z}_0^{\perp} \subset \mathfrak{z}$  be the orthogonal spaces of  $\mathfrak{p}_0, \mathfrak{k}_0, \mathfrak{z}_0$  with respect to B. Let  $\mathfrak{p}_0^{\perp}(\gamma) \subset \mathfrak{p}_0$ ,  $\mathfrak{k}_0^{\perp}(\gamma) \subset \mathfrak{k}_0$ ,  $\mathfrak{z}_0^{\perp}(\gamma) \subset \mathfrak{z}_0$  be the orthogonal spaces of  $\mathfrak{p}(\gamma), \mathfrak{k}(\gamma), \mathfrak{z}(\gamma)$  with respect to B. Then

$$\mathfrak{z}_0^{\perp} = \mathfrak{p}_0^{\perp} \oplus \mathfrak{k}_0^{\perp}, \qquad \qquad \mathfrak{z}_0^{\perp}(\gamma) = \mathfrak{p}_0^{\perp}(\gamma) \oplus \mathfrak{k}_0^{\perp}(\gamma).$$
 (5.69)

In particular, we have the orthogonal decompositions with respect to B,

$$\mathfrak{p} = \underbrace{\mathfrak{p}(\gamma) \oplus \mathfrak{p}_0^{\perp}(\gamma)}_{\mathfrak{p}_0} \oplus \mathfrak{p}_0^{\perp}, \quad \mathfrak{k} = \underbrace{\mathfrak{k}(\gamma) \oplus \mathfrak{k}_0^{\perp}(\gamma)}_{\mathfrak{k}_0} \oplus \mathfrak{k}_0^{\perp}, \quad \mathfrak{g} = \underbrace{\mathfrak{z}(\gamma) \oplus \mathfrak{z}_0^{\perp}(\gamma)}_{\mathfrak{k}_0} \oplus \mathfrak{z}_0^{\perp}. \quad (5.70)$$

Let us introduce the *J*-function [B11, Section 5.5] of Bismut. In [B11, Section 5.5], it has been shown that the analytic function defined for  $Y \in \mathfrak{k}(\gamma)$  by

$$\frac{1}{\det\left(1 - \operatorname{Ad}(k^{-1})\right)|_{\mathfrak{z}_{0}^{\perp}(\gamma)}} \frac{\det\left(1 - \exp(-i\operatorname{ad}(Y))\operatorname{Ad}(k^{-1})\right)|_{\mathfrak{t}_{0}^{\perp}(\gamma)}}{\det\left(1 - \exp(-i\operatorname{ad}(Y))\operatorname{Ad}(k^{-1})\right)|_{\mathfrak{p}_{0}^{\perp}(\gamma)}}$$
(5.71)

has a natural square root, which is still analytic in  $Y \in \mathfrak{k}(\gamma)$ . For a Hermitian matrix H, define

$$\widehat{A}(H) = \det^{1/2} \left( \frac{H/2}{\sinh(H/2)} \right). \tag{5.72}$$

The square root in (5.72) is the positive square root of a positive real number.

**Definition 5.16.** Let  $J_{\gamma}$  be the analytic function on  $\mathfrak{k}(\gamma)$  defined for  $Y \in \mathfrak{k}(\gamma)$  by

$$J_{\gamma}(Y) = \frac{1}{\left|\det\left(1 - \operatorname{Ad}(\gamma)\right)\right|_{\mathfrak{z}_{0}^{\perp}}\left|^{1/2}} \frac{\widehat{A}\left(i\operatorname{ad}(Y)|_{\mathfrak{p}(\gamma)}\right)}{\widehat{A}\left(i\operatorname{ad}(Y)|_{\mathfrak{k}(\gamma)}\right)}$$

$$\left[\frac{1}{\det\left(1 - \operatorname{Ad}(k^{-1})\right)|_{\mathfrak{z}_{0}^{\perp}(\gamma)}} \frac{\det\left(1 - \exp(-i\operatorname{ad}(Y))\operatorname{Ad}(k^{-1})\right)|_{\mathfrak{k}_{0}^{\perp}(\gamma)}}{\det\left(1 - \exp(-i\operatorname{ad}(Y))\operatorname{Ad}(k^{-1})\right)|_{\mathfrak{p}_{0}^{\perp}(\gamma)}}\right]^{1/2}. (5.73)$$

**Remark 5.17.** There is  $C_{\gamma} > 0$  such that for all  $Y \in \mathfrak{k}(\gamma)$ , we have

$$|J_{\gamma}(Y)| \leqslant C_{\gamma} e^{C_{\gamma}|Y|}. (5.74)$$

Recall that  $C^{\mathfrak{k},\mathfrak{p}}$  and  $C^{\mathfrak{k},\mathfrak{k}}$  are defined as in (5.50) associated with the adjoint actions of K on  $\mathfrak{p}$  and  $\mathfrak{k}$ . Write

$$c_{\mathfrak{g}} = -\frac{1}{8} \operatorname{Tr}^{\mathfrak{p}} [C^{\mathfrak{k},\mathfrak{p}}] - \frac{1}{24} \operatorname{Tr}^{\mathfrak{k}} [C^{\mathfrak{k},\mathfrak{k}}] \in \mathbf{R}_{+}. \tag{5.75}$$

For  $Y \in \mathfrak{k}(\gamma)$ , let dY be the Lebesgue measure on  $\mathfrak{k}(\gamma)$  induced by -B. The main result of [B11, Theorem 6.1.1] is the following.

**Theorem 5.18.** For t > 0, we have

$$\operatorname{Tr}^{[\gamma]}\left[\exp\left(-tC^{\mathfrak{g},X,\tau}/2\right)\right] = \frac{1}{(2\pi t)^{\dim \mathfrak{z}(\gamma)/2}} \exp\left(-\frac{|a|^2}{2t} - \frac{c_{\mathfrak{g}}}{2}t\right)$$
$$\int_{Y\in\mathfrak{k}(\gamma)} J_{\gamma}(Y) \operatorname{Tr}^{E_{\tau}}\left[\tau\left(k^{-1}\right) \exp(-i\tau(Y))\right] \exp\left(-|Y|^2/2t\right) dY. \quad (5.76)$$

### 5.6. Moscovici-Stanton's vanishing theorem

In this section, we show Theorem 5.9 in the case of  $\delta(G) \neq 1$ . By Remark 5.13, it is enough to show  $T_F(Z) = 1$ .

Let  $\mathfrak{t} \subset \mathfrak{k}$  be a Cartan subalgebra of  $\mathfrak{k}$ . Set

$$\mathfrak{b} = \{ Y \in \mathfrak{p} : [Y, \mathfrak{t}] = 0 \}. \tag{5.77}$$

Then  $\mathfrak{h} = \mathfrak{b} \oplus \mathfrak{t}$  is a fundamental Cartan subalgebra of  $\mathfrak{g}$ .

Let  $\gamma = e^a k^{-1} \in G$  be a semisimple element such that (5.36) holds. Let  $\mathfrak{t}(\gamma) \subset \mathfrak{k}(\gamma)$  be a Cartan subalgebra of  $\mathfrak{k}(\gamma)$ . Set

$$\mathfrak{b}(\gamma) = \{ Y \in \mathfrak{p} : [Y, \mathfrak{t}(\gamma)] = 0, \operatorname{Ad}(k)Y = Y \}. \tag{5.78}$$

By (5.77) and (5.78), we have

$$\dim \mathfrak{b}(\gamma) \geqslant \dim \mathfrak{b} = \delta(G). \tag{5.79}$$

The following theorem is [B11, Theorem 7.9.1] (see also [Sh18, Theorem 4.12]).

**Theorem 5.19.** Let  $\gamma \in G$  be semisimple such that  $\dim \mathfrak{b}(\gamma) \geqslant 2$ . For  $Y \in \mathfrak{k}(\gamma)$ , we have

$$\operatorname{Tr}_{\mathrm{s}}^{\Lambda^{\cdot}(\mathfrak{p}^{*})} \left[ N^{\Lambda^{\cdot}(\mathfrak{p}^{*})} \operatorname{Ad}(k^{-1}) \exp(-i \operatorname{ad}(Y)) \right] = 0. \tag{5.80}$$

In particular, for t > 0, we have

$$\operatorname{Tr}_{\mathrm{s}}^{[\gamma]} \left[ N^{\Lambda^{\cdot}(T^*X)} \exp\left(-tC^{\mathfrak{g},X}/2\right) \right] = 0. \tag{5.81}$$

*Proof.* Since the term on the left-hand side of (5.80) is Ad  $(K(\gamma))$ -invariant, it is enough to show (5.80) for  $Y \in \mathfrak{t}(\gamma)$ . If  $Y \in \mathfrak{t}(\gamma)$ , we have

$$\operatorname{Tr}_{\mathbf{s}}^{\Lambda^{+}(\mathfrak{p}^{*})} \left[ N^{\Lambda^{+}(\mathfrak{p}^{*})} \operatorname{Ad}(k^{-1}) \exp(-i \operatorname{ad}(Y)) \right] \\
= \frac{\partial}{\partial b} \Big|_{b=0} \det \left( 1 - e^{b} \operatorname{Ad}(k) \exp(i \operatorname{ad}(Y)) \right) \Big|_{\mathfrak{p}}.$$
(5.82)

Since dim  $\mathfrak{b}(\gamma) \ge 2$ , by (5.82), we get (5.80) for  $Y \in \mathfrak{t}(\gamma)$ . By (5.76) and (5.80), we get (5.81).

In this way, Bismut [B11, Theorem 7.9.3] recovered [MoSt91, Corollary 2.2].

**Corollary 5.20.** Let F be a unitarily flat vector bundle on Z. Assume that dim Z is odd and  $\delta(G) \neq 1$ . Then for any t > 0, we have

$$\operatorname{Tr}_{\mathbf{s}} \left[ N^{\Lambda^{*}(T^{*}Z)} \exp\left(-t\Box^{Z}/2\right) \right] = 0. \tag{5.83}$$

In particular,

$$T_F(Z) = 1.$$
 (5.84)

*Proof.* Since dim Z and  $\delta(G)$  have the same parity, and since dim Z is odd and  $\delta(G) \neq 1$ , we get  $\delta(G) \geq 3$ . By (5.79), dim  $\mathfrak{b}(\gamma) \geq 3$ . Thus, (5.83) is a consequence of (5.52), (5.61), and (5.81).

# 5.7. The case $G = SO^0(p, 1)$ with p odd

In this section, we assume  $G = SO^0(p, 1)$  with p odd. Then G is a semisimple Lie group of **R**-rank 1, and X = G/K is hyperbolic space of dimension p. The result of this section is due to Fried [F86a].

It is easy to show that up to a sign there exists  $\alpha \in \mathfrak{b}^*$  such that for  $a \in \mathfrak{b}$ ,  $\mathrm{ad}(a)$  acting on  $\mathfrak{g}$  has three eigenvalues  $0, \pm \langle \alpha, a \rangle$ . We have an orthogonal splitting with respect to B,

$$\mathfrak{g} = \mathfrak{b} \oplus \mathfrak{m} \oplus \mathfrak{n} \oplus \overline{\mathfrak{n}}, \tag{5.85}$$

where  $\mathfrak{b} \oplus \mathfrak{m}$  (resp.  $\mathfrak{n}$ , resp.  $\overline{\mathfrak{n}}$ ) is the eigenspace of  $\mathrm{ad}(a)$  for the eigenvalue 0 (resp.  $\langle \alpha, a \rangle$ , resp.  $-\langle \alpha, a \rangle$ ). Let  $\mathcal{M} \subset G$  be the connected subgroup of G associated to  $\mathfrak{m}$ . Then  $\mathcal{M}$  is compact and isomorphic to  $\mathrm{SO}(p-1)$ . Moreover, the action of  $\mathcal{M}$  on  $\mathfrak{n}$  is just the  $\mathrm{SO}(p-1)$ -action on  $\mathbb{R}^{p-1}$ .

One feature of the group  $SO^0(p, 1)$  with p odd is that any non-elliptic semisimple element of  $\gamma$  can be conjugate to  $e^a k^{-1}$  such that  $a \in \mathfrak{b}, a \neq 0$  and  $k \in \mathcal{M}$ . Moreover, the subspaces  $\mathfrak{z}_0$  and  $\mathfrak{z}_0^{\perp}$  introduced in (5.68) and (5.69) do not depend on the choice of non-elliptic semisimple element of  $\gamma$ , and is given by

$$\mathfrak{z}_0 = \mathfrak{b} \oplus \mathfrak{m}, \qquad \qquad \mathfrak{z}_0^{\perp} = \mathfrak{n} \oplus \overline{\mathfrak{n}}.$$
 (5.86)

**Definition 5.21.** For  $\sigma \in \mathbf{C}$ , we define a formal Selberg zeta function associated to a representation  $\eta$  of  $\mathcal{M}$  and to a representation  $\rho$  of  $\Gamma$  by

$$Z_{\eta,\rho}(\sigma) = \exp\left(-\sum_{\substack{[\gamma] \in [\Gamma] - \{1\} \\ \gamma \sim e^a k^{-1}}} \operatorname{Tr}[\rho(\gamma)] \frac{\operatorname{Tr}^{E_{\eta}} \left[\eta(k^{-1})\right]}{\left|\det\left(1 - \operatorname{Ad}(\gamma)\right)\right|_{\mathfrak{z}_0^{\perp}}\right|^{1/2}} \frac{e^{-\sigma\ell_{[\gamma]}}}{m_{[\gamma]}}\right). \tag{5.87}$$

The formal Selberg zeta function is said to be well defined if the same conditions as in Definition 2.7 hold.

For  $\gamma = e^a k^{-1}$  such that  $a \in \mathfrak{b}, a \neq 0$  and  $k \in \mathcal{M}$ , we have

$$\left| \det \left( 1 - \operatorname{Ad}(\gamma) \right) \right|_{\mathfrak{z}_{0}^{\perp}} \right|^{1/2} = \sum_{j=0}^{\dim \mathfrak{n}} (-1)^{j} \operatorname{Tr}^{\Lambda^{j}(\mathfrak{n}^{*})} \left[ \operatorname{Ad} \left( k^{-1} \right) \right] e^{\left( \frac{1}{2} \dim \mathfrak{n} - j \right) |\alpha| |a|}. \quad (5.88)$$

If we consider the Selberg zeta function associated to  $\eta_j = \Lambda^j(\mathfrak{n}^*)$ , using (5.88) and  $\ell_{[\gamma]} = |a|$ , we have

$$R_{\phi,\rho}(\sigma) = \prod_{j=0}^{\dim \mathfrak{n}} Z_{\eta_j,\rho} \left( \sigma + \left( j - \frac{\dim \mathfrak{n}}{2} \right) |\alpha| \right)^{(-1)^{j-1}}. \tag{5.89}$$

To show the meromorphic extension of  $R_{\phi,\rho}$ , it is enough to consider the meromorphic extension of  $Z_{\eta_i,\rho}$ .

There are two remarkable properties of  $\eta_j$ . First, since  $\eta_j$  is just the representation of SO(p-1) on  $\Lambda^j(\mathbf{R}^{p-1})$ , the Casimir of  $\mathfrak{m}$  acts on  $\eta_j$  is a scalar. Second,  $\eta_j$  has a lift as a virtual representation of K. More precisely, let RO(K),  $RO(\mathcal{M})$  be the real representation rings of K and  $\mathcal{M}$ . The restriction induces a morphism of rings

$$\iota: RO(K) \to RO(\mathcal{M}).$$
 (5.90)

Since K and  $\mathcal{M}$  have the same complex rank,  $\iota$  is an injection. The second property is that  $\eta_j$  has a lift  $\widehat{\eta}_j = \widehat{\eta}_j^+ - \widehat{\eta}_j^- \in RO(K)$ , where

$$\widehat{\eta}_{j}^{+} = \Lambda^{j}(\mathfrak{p}^{*}) + \Lambda^{j-2}(\mathfrak{p}^{*}) + \cdots, \quad \widehat{\eta}_{j}^{-} = \Lambda^{j-1}(\mathfrak{p}^{*}) + \Lambda^{j-3}(\mathfrak{p}^{*}) + \cdots.$$
 (5.91)

For a virtual representation  $\widehat{\eta} = \widehat{\eta}^+ - \widehat{\eta}^- \in RO(K)$ , where  $\widehat{\eta}^+$ ,  $\widehat{\eta}^-$  are two representations of K, we use the notation

$$\det_{\mathbf{s}} \left( \sigma + C^{\mathfrak{g}, Z, \widehat{\eta}, \rho} \right) = \frac{\det \left( \sigma + C^{\mathfrak{g}, Z, \widehat{\eta}^+, \rho} \right)}{\det \left( \sigma + C^{\mathfrak{g}, Z, \widehat{\eta}^-, \rho} \right)}. \tag{5.92}$$

Recall that  $c_{\mathfrak{g}} \in \mathbf{R}_{+}$  is defined in (5.75). We define  $c_{\mathfrak{m}} \in \mathbf{R}_{+}$  in the same way.

**Proposition 5.22.** Assume that  $\eta$  has a lift  $\widehat{\eta} \in RO(K)$  and that the Casimir of  $\mathfrak{m}$  on  $E_{\eta}$  is a scalar  $C^{\mathfrak{m},\eta} \in \mathbf{R}_{-}$ . Then  $Z_{\eta,\rho}(\sigma)$  is well defined such that

$$Z_{\eta,\rho}(\sigma) = e^{P_{\eta}(\sigma)} \det_{\mathbf{s}} \left( \sigma^2 - c_{\mathfrak{g}} + c_{\mathfrak{m}} - C^{\mathfrak{m},\eta} + C^{\mathfrak{g},Z,\widehat{\eta},\rho} \right). \tag{5.93}$$

where  $P_{\eta}(\sigma)$  is an odd polynomial of  $\sigma$ .

*Proof.* By a general theorem of elliptic operators [Vo87], we know that the right-hand side of (5.93) has a meromorphic extension to  $\mathbf{C}$ . So it is enough to show (5.93) for  $\sigma \in \mathbf{R}$  and  $\sigma \gg 1$ . In this case, using the Mellin transform, we have

$$\det_{\mathbf{s}} \left( \sigma^2 - c_{\mathfrak{g}} + c_{\mathfrak{m}} - C^{\mathfrak{m}, \eta} + C^{\mathfrak{g}, Z, \widehat{\eta}, \rho} \right) \tag{5.94}$$

$$= \exp\left(-\frac{\partial}{\partial s}\Big|_{s=0} \frac{1}{\Gamma(s)} \int_0^\infty \mathrm{Tr}_{\mathbf{s}} \left[ \exp\left(-t \left(\sigma^2 - c_{\mathfrak{g}} + c_{\mathfrak{m}} - C^{\mathfrak{m},\eta} + C^{\mathfrak{g},Z,\widehat{\eta},\rho}\right)\right) \right] t^{s-1} dt \right).$$

We use the Selberg trace formula to evaluate the second line of (5.94). For the group of **R**-rank 1, Bismut's orbital integral formula has a very simple form [B11, Theorem 8.2.1]. For  $\gamma = e^a k^{-1}$  with  $a \in \mathfrak{b}$ ,  $a \neq 0$ , and  $k \in \mathcal{M}$ , we have

$$\operatorname{Tr}^{[\gamma]}\left[\exp\left(-tC^{\mathfrak{g},X,\widehat{\eta}}/2\right)\right]$$
 (5.95)

$$=\frac{1}{\left|\det\left(1-\mathrm{Ad}(\gamma)\right)|_{\mathfrak{z}_{0}^{\perp}}\right|^{1/2}}\frac{1}{\sqrt{2\pi t}}\exp\left(-\frac{|a|^{2}}{2t}+\frac{-c_{\mathfrak{g}}+c_{\mathfrak{m}}-C^{\mathfrak{m},\eta}}{2}t\right)\mathrm{Tr}\left[\eta(k^{-1})\right],$$

and if  $\gamma = 1$ , we have

$$\operatorname{Tr}^{[\gamma]}\left[\exp\left(-tC^{\mathfrak{g},X,\widehat{\eta}}/2\right)\right] = \frac{1}{\sqrt{t}}Q_{\eta}\left(\frac{1}{t}\right)\exp\left(\frac{-c_{\mathfrak{g}} + c_{\mathfrak{m}} - C^{\mathfrak{m},\eta}}{2}t\right). \tag{5.96}$$

where  $Q_{\eta}$  is some explicit polynomial. Using Theorem 5.15, (5.94)–(5.96), we deduce (5.93) for  $\sigma \in \mathbf{R}$  and  $\sigma \gg 1$ .

Let us give an explicit formula for the constant appearing in Theorem 5.9. For  $0 \le j \le \dim \mathfrak{n}$ , set

$$r_{i} = \dim \ker C^{\mathfrak{g}, Z, \widehat{\eta}_{i}^{+}, \rho} - \dim \ker C^{\mathfrak{g}, Z, \widehat{\eta}_{i}^{-}, \rho}.$$
 (5.97)

By (5.52) and (5.91), we have

$$r_j = b_j(Z, F) - b_{j-1}(Z, F) + b_{j-2}(Z, F) - \cdots,$$
 (5.98)

where  $b_j(Z, F) = \dim H^j(Z, F)$  is the Betti number. Clearly,  $r_j = r_{\dim \mathfrak{n} - j}$ .

**Definition 5.23.** Set

$$C_{\rho} = \prod_{j=0}^{\frac{\dim \mathfrak{n}}{2} - 1} \left( -4 \left( \frac{\dim \mathfrak{n}}{2} - j \right)^{2} |\alpha|^{2} \right)^{(-1)^{j-1} r_{j}}, \quad r_{\rho} = 2 \sum_{j=0}^{\frac{\dim \mathfrak{n}}{2}} (-1)^{j-1} r_{j}. \quad (5.99)$$

Let us complete the proof of Theorem 5.9 in the case  $G = SO^0(p, 1)$  with p odd.

Proof of (5.43) and (5.44). Simple calculation shows that

$$c_{\mathfrak{g}} - c_{\mathfrak{m}} + C^{\mathfrak{m}, \eta_j} = \left| j - \frac{\dim \mathfrak{n}}{2} \right|^2 |\alpha|^2. \tag{5.100}$$

By (5.89), (5.93), and (5.100), using the fact that  $P_{\eta}$  is odd, we see that as  $\sigma \to 0$ ,

$$R_{\phi,\rho}(\sigma) = \left(1 + \mathcal{O}(\sigma)\right) \prod_{j=0}^{\dim \mathfrak{n}} \det_{\mathbf{s}} \left(\sigma^2 + 2\sigma \left(j - \frac{\dim \mathfrak{n}}{2}\right) |\alpha| + C^{\mathfrak{g},Z,\widehat{\eta}_j,\rho}\right)^{(-1)^{j-1}}.$$
(5.101)

On the other hand, by (1.8), (5.52), and (5.91), we have

$$\prod_{j=0}^{\dim \mathfrak{n}} \det_{\mathbf{s}} \left( \sigma + C^{\mathfrak{g}, Z, \widehat{\eta}_{j}, \rho} \right)^{(-1)^{j-1}} = \prod_{q=1}^{\dim Z} \det \left( \sigma + \square_{q}^{Z} \right)^{(-1)^{q} q}$$

$$= \left( 1 + \mathcal{O}(\sigma) \right) \sigma^{\sum_{q=1}^{\dim Z} (-1)^{q} q b_{q}(Z, F)} T_{F}(Z)^{2}.$$
(5.102)

Comparing (5.101) and (5.102), by (5.98), we get (5.43). By (5.98), if  $H^{\cdot}(Z, F) = 0$ , then (5.44) follows.

## 5.8. The reductive group G with $\delta(G) = 1$

In this section, we assume that G is a real reductive group with  $\delta(G) = 1$ . From Table 1, we see three examples  $\mathbf{R}$ ,  $\mathrm{SL}_3(\mathbf{R})$ , and  $\mathrm{SO}^0(p,q)$  with pq > 1 odd. The result in this section is from [Sh18].

**5.8.1. Meromorphic extension.** We generalize the construction in Section 5.7 to the case in this section. By [Sh18, Proposition 6.2], we can define  $\mathfrak{m}, \mathfrak{n}, \overline{\mathfrak{n}}$  is the same way. So the splitting (5.85) still holds true, and dim  $\mathfrak{n}$  is still even. However, the group  $\mathcal{M}$  is not necessarily compact. It is a reductive group with maximal compact subgroup  $K \cap \mathcal{M}$  and with Cartan decomposition

$$\mathfrak{m} = \mathfrak{p}_{\mathfrak{m}} \oplus \mathfrak{k}_{\mathfrak{m}}. \tag{5.103}$$

One of difference with the **R**-rank 1 case is that a general non-elliptic semisimple element can not always be conjugate to  $e^a k^{-1}$  with  $a \in \mathfrak{b}$ ,  $a \neq 0$ , and  $k \in \mathcal{M} \cap K$ . In [Sh18, Proposition 4.1], we observe that this happens if and only if

$$\dim \mathfrak{b}(\gamma) = 1. \tag{5.104}$$

For such an element  $\gamma$ , we still have (5.86). Moreover, if dim  $\mathfrak{b}(\gamma) \geq 2$ , as in Remark 5.13, we have

$$\chi_{\text{orb}}(B_{[\gamma]}/\mathbb{S}^1) = 0. \tag{5.105}$$

This gives the motivation to introduce the following Selberg zeta function.

**Definition 5.24.** For  $\sigma \in \mathbf{C}$ , we define a formal Selberg zeta function associated to a representation  $\eta$  of  $\mathcal{M}$  and to a representation  $\rho$  of  $\Gamma$  by

$$Z_{\eta,\rho}(\sigma) = \exp\left(-\sum_{\substack{[\gamma] \in [\Gamma] - \{1\} \\ \gamma \sim e^{a_k} - 1}} \operatorname{Tr}[\rho(\gamma)] \frac{\operatorname{Tr}^{E_{\eta}} \left[\eta(k^{-1})\right]}{\left|\det\left(1 - \operatorname{Ad}(\gamma)\right)\right|_{\mathfrak{z}_{0}^{\perp}}\right|^{1/2}} \frac{\chi_{\operatorname{orb}}(B_{[\gamma]}/\mathbb{S}^{1})}{m_{[\gamma]}} e^{-\sigma\ell_{[\gamma]}}\right).$$

$$(5.106)$$

Thanks to (5.105), the semisimple element  $\gamma$  satisfying dim  $\mathfrak{b}(\gamma) \geq 2$  does not have a contribution to the Selberg zeta function. Take  $\eta_j = \Lambda^j(\mathfrak{n}^*)$  as before. Equation (5.89) still holds true. So we can reduce our problem to showing the meromorphic extension of  $Z_{\eta_j,\rho}$ .

Note that in this case, the Casimir operator of  $\mathfrak{m}$  still acts as a constant on  $\eta_j$ . For technical reasons, we note also that the compact dual of  $\mathcal{M}$  acts on  $\eta_j \otimes_{\mathbf{R}} \mathbf{C}$ . Also, the lifting property still holds but in a more complicated form. Consider the diagram

$$\operatorname{RO}(\mathcal{M})$$

$$\downarrow^{\iota_{\mathcal{M}}}$$
 $\operatorname{RO}(K) \xrightarrow{\iota_{K}} \operatorname{RO}(K \cap \mathcal{M}),$ 

$$(5.107)$$

where the maps are induced by restriction. In [Sh18, Theorem 6.11], we show that  $\iota_{\mathcal{M}}(\eta_j)$ ,  $\Lambda^j(\mathfrak{p}_{\mathfrak{m}}^*) \in \mathrm{RO}(K \cap \mathcal{M})$  have unique lifts in  $\mathrm{RO}(K)$ .

**Proposition 5.25.** Suppose that  $(\eta, E_{\eta})$  is a finite-dimensional real representation of  $\mathcal{M}$  such that

- the Casimir of  $\mathfrak{m}$  acting on  $E_{\eta}$  is a scalar  $C^{\mathfrak{m},\eta} \in \mathbf{R}_{-}$ ;
- the compact dual of  $\mathcal{M}$  acts on  $E_{\eta} \otimes_{\mathbf{R}} \mathbf{C}$ ;
- the restriction  $\iota_{\mathcal{M}}(\eta)$  to  $K \cap \mathcal{M}$  has a lift in RO(K).

Then  $Z_{n,\rho}(\sigma)$  is well defined such that

$$Z_{\eta,\rho}(\sigma) = e^{P_{\eta}(\sigma)} \det_{\mathbf{s}} \left( \sigma^2 - c_{\mathfrak{g}} + c_{\mathfrak{m}} - C^{\mathfrak{m},\eta} + C^{\mathfrak{g},Z,\widehat{\eta},\rho} \right), \tag{5.108}$$

where  $P_{\eta}(\sigma)$  is an odd polynomial of  $\sigma$ , and  $\widehat{\eta} \in RO(K)$  is a virtual representation of K such that

$$\iota_K(\widehat{\eta}) = \Lambda^{\cdot}(\mathfrak{p}_{\mathfrak{m}}^*) \otimes \iota_{\mathcal{M}}(\eta) \in RO(K \cap \mathcal{M}).$$
 (5.109)

*Proof.* The proof of our proposition is similar to the one of Proposition 5.22, except that the evaluation of the orbital integral is more complicated. Note that when  $\dim \mathfrak{b}(\gamma) \geq 2$ , we have

$$\operatorname{Tr}^{[\gamma]} \left[ \exp \left( -tC^{\mathfrak{g}, X, \widehat{\eta}} / 2 \right) \right] = 0. \tag{5.110}$$

This is a refinement of Theorem 5.19.

Now a method similar to the one given in Section 5.7 shows (5.43) with the constants  $C_{\rho}$  and  $r_{\rho}$  defined by the same formula as in (5.99).

**5.8.2. Regularity at**  $\sigma = 0$ . The proof of (5.44) is much more difficult since we do not have a relation between  $r_j$  and  $b_j(Z, F)$  as in (5.98). We rely on some deep results from the representation theory. Here we only sketch the main steps.

Let  $\widehat{G}_u$  be the unitary dual of G. For a unitary representation  $V_{\pi} \in \widehat{G}_u$ , denote by  $V_{\pi,K}$  the associated Harish-Chandra  $(\mathfrak{g},K)$ -module, which is formed by K-finite elements. The center of the complexified enveloping algebra  $\mathscr{U}(\mathfrak{g}_{\mathbf{C}})$  acts on  $V_{\pi,K}$  as scalars, which is called the infinitesimal character and is denoted by  $\chi_{\pi}$ . Clearly, for  $a \in \mathbf{C}$ , we have

$$\chi_{\pi}(a) = a. \tag{5.111}$$

We call  $\chi_{\pi}$  is trivial if  $\chi_{\pi}$  coincides with the infinitesimal character of the trivial representation of G.

Recall that  $\widehat{p}: \Gamma \backslash G \to Z$  is the natural projection. The group G acts unitarily on the right on  $L^2(\Gamma \backslash G, \widehat{p}^*F)$ . By [GeGraPS69, p.23, Theorem],  $L^2(\Gamma \backslash G, \widehat{p}^*F)$  decomposes into a discrete Hilbert direct sum with finite multiplicity of unitary representations of G. We can write

$$L^{2}(\Gamma \backslash G, \widehat{p}^{*}F) = \bigoplus_{\pi \in \widehat{G}_{n}}^{\text{Hil}} n_{\rho}(\pi)V_{\pi}, \qquad (5.112)$$

with  $n_{\rho}(\pi) < \infty$ .

Recall that  $\tau$  is a real finite-dimensional orthogonal representation of K on the real Euclidean space  $E_{\tau}$ , and that  $C^{\mathfrak{g},Z,\tau,\rho}$  is the Casimir element of G acting on  $C^{\infty}(Z,\mathcal{F}_{\tau}\otimes_{\mathbf{C}}F)$ . By (5.112), we have

$$\ker C^{\mathfrak{g},Z,\tau,\rho} = \bigoplus_{\pi \in \widehat{G}_u, \chi_{\pi}(C^{\mathfrak{g}}) = 0} n_{\rho}(\pi) (V_{\pi,K} \otimes_{\mathbf{R}} E_{\tau})^K, \tag{5.113}$$

By properties of elliptic operators, the sum on right-hand side of (5.113) is finite. We will apply (5.113) in the case  $\tau = \Lambda^j(\mathfrak{p}^*)$  and also  $\tau = \widehat{\eta}_j$ .

The case  $\tau = \Lambda^{j}(\mathfrak{p}^{*})$ . By (5.52), (5.113), and by the Hodge theory (1.5), we have

$$H^{\cdot}(Z,F) = \bigoplus_{\pi \in \widehat{G}_{u,\chi_{\pi}(C^{\mathfrak{g}})=0}} n_{\rho}(\pi) \left(V_{\pi,K} \otimes_{\mathbf{R}} \Lambda^{\cdot}(\mathfrak{p}^{*})\right)^{K}.$$
 (5.114)

The Hodge theory for Lie algebras [BorW00, Proposition II.3.1] tells us that the right-hand side of (5.114) has a cohomological interpretation,

$$H^{\cdot}(Z,F) = \bigoplus_{\pi \in \widehat{G}_{n}, \chi_{\pi}(C^{\mathfrak{g}}) = 0} n_{\rho}(\pi) H^{\cdot}(\mathfrak{g}, K; V_{\pi,K}), \tag{5.115}$$

where  $H^{\cdot}(\mathfrak{g}, K; V_{\pi,K})$  is the  $(\mathfrak{g}, K)$ -cohomology of the Harish-Chandra  $(\mathfrak{g}, K)$ -module  $V_{\pi,K}$ . By the following property of  $(\mathfrak{g}, K)$ -cohomology [VZu84, V84, SR99] (see also [Sh18, Theorem 8.9]), for  $\pi \in \widehat{G}_u$ ,

$$\chi_{\pi}$$
 is trivial  $\iff H^{\cdot}(\mathfrak{g}, K; V_{\pi, K}) \neq 0,$  (5.116)

we see that the sum in (5.115) can be reduced to  $\pi \in \widehat{G}_u$  with trivial infinitesimal character,

$$H^{\cdot}(Z,F) = \bigoplus_{\pi \in \widehat{G}_{u,\chi_{\pi} \text{ trivial}}} n_{\rho}(\pi) H^{\cdot}(\mathfrak{g},K;V_{\pi,K}), \tag{5.117}$$

and each summand does not vanish except for  $n_{\rho}(\pi) = 0$ .

The case  $\tau = \hat{\eta}_j$ . By (5.113), we have

$$r_{j} = \sum_{\pi \in \widehat{G}_{n}, \chi_{\pi}(C^{\mathfrak{g}}) = 0} n_{\rho}(\pi) \left( \dim \left( V_{\pi,K} \otimes_{\mathbf{R}} \widehat{\eta}_{j}^{+} \right)^{K} - \dim \left( V_{\pi,K} \otimes_{\mathbf{R}} \widehat{\eta}_{j}^{-} \right)^{K} \right). \tag{5.118}$$

As (5.115), in [Sh18, Theorem 8.14, Corollary 8.15], we give a cohomology interpretation of the right-hand side of (5.118),

$$r_{j} = \frac{1}{\chi(K/K \cap \mathcal{M})} \sum_{\pi \in \widehat{G}_{u}, \chi_{\pi}(C^{\mathfrak{g}}) = 0} n_{\rho}(\pi)$$

$$\times \sum_{i=0}^{\dim \mathfrak{p}_{\mathfrak{m}}} \sum_{j=0}^{\dim \mathfrak{n}} (-1)^{i+j} \dim H^{i}(\mathfrak{m}, K \cap \mathcal{M}; H_{j}(\mathfrak{n}, V_{\pi,K}) \otimes_{\mathbf{R}} E_{\eta}), \quad (5.119)$$

where  $H_j(\mathfrak{n}, V_{\pi,K})$  is the  $\mathfrak{n}$ -homology of  $V_{\pi,K}$  and is a  $(\mathfrak{m}, K \cap \mathcal{M})$ -module. Moreover, as (5.117), [Sh18, Proposition 8.17, Corollary 8.18] implies that the first sum in (5.119) can be reduced to  $\pi \in \widehat{G}_u$  with trivial infinitesimal character, i.e.,

$$r_{j} = \frac{1}{\chi(K/K \cap \mathcal{M})} \sum_{\pi \in \widehat{G}_{u}, \chi_{\pi} \text{ trivial}} n_{\rho}(\pi)$$

$$\times \sum_{i=0}^{\dim \mathfrak{p}_{\mathfrak{m}} \dim \mathfrak{n}} (-1)^{i+j} \dim H^{i}(\mathfrak{m}, K \cap \mathcal{M}; H_{j}(\mathfrak{n}, V_{\pi,K}) \otimes_{\mathbf{R}} E_{\eta}).$$
(5.120)

Equations (5.117) and (5.120) can be considered as an analogue of (5.98). Now we prove (5.44).

The proof of (5.44). If  $H^{\cdot}(Z, F) = 0$ , by (5.117), we see that if  $\chi_{\pi}$  is trivial, then  $n_{\rho}(\pi) = 0$ . By (5.120), we see that  $r_{j} = 0$  for all j. By (5.99), we complete the proof of (5.44).

## 6. Anosov flow

The purpose of this section is to study the Fried conjecture for the Anosov flow. This section is organized as follows. In Section 6.1, we introduce the Anosov flow.

In Section 6.2, we explain the meromorphic extension of the Ruelle dynamical zeta function [GiLiPo13, DyZ16].

In Section 6.3, we explain a proof that under certain resonance conditions the value at zero of the Ruelle dynamical zeta function does not depend on a small perturbation of the Anosov flow.

In Section 6.4, we study the Anosov flow on 3-manifolds.

#### 6.1. Closed orbits of the Anosov flow

Let M be a closed manifold with a smooth vector field  $V \in C^{\infty}(M, TM)$ . Let  $(\phi_t)_{t \in \mathbf{R}}$  be the flow on M generated by V.

**Definition 6.1.** A flow  $\phi$  is called Anosov if there is a  $\phi_t$ -invariant continuous splitting  $^{12}$ 

$$TM = \mathbf{R}V \oplus E^u \oplus E^s \tag{6.1}$$

of  $C^0$ -vector bundles on M and there exist  $C > 0, \theta > 0$  and a Riemannian metric on M such that for  $v \in E_x^u$ ,  $v' \in E_x^s$ , and t > 0, we have

$$|\phi_{-t,*}v| \le Ce^{-\theta|t|}|v|, \qquad |\phi_{t,*}v'| \le Ce^{-\theta|t|}|v'|.$$
 (6.2)

<sup>&</sup>lt;sup>12</sup>This requires that  $\mathbf{R}V$  is a line bundle on M. It implies  $V(x) \neq 0$  for all  $x \in M$ , and so the Euler characteristic number  $\chi(M)$  vanishes.

In this section, we assume  $\phi$ . is Anosov. It is well known that the set of closed orbits  $\overline{\wp}(\phi)$  defined in (2.12) is discrete (see [Mar04] or [DyZ16, Appendix A]), so that Assumption 2.4 holds. More precisely, for  $\gamma \in \overline{\wp}(\phi)$ , let  $\ell_{\gamma} \in \mathbf{R}_{+}^{*}$ ,  $m_{\gamma} \in \mathbf{N}^{*}$  be the period and the multiplicity of  $\gamma$ . Then (2.14) becomes

$$\wp(\phi.) = \coprod_{\gamma \in \overline{\wp}(\phi.)} \mathbb{S}^1 \times \{\ell_{\gamma}\}. \tag{6.3}$$

Moreover, by [Mar04, Theorem 1.1, p.78] or [DyZ16, Lemma 2.2] there is C > 0 such that for  $r \ge 0$ , we have

$$|\{\gamma \in \overline{\wp}(\phi.) : \ell_{\gamma} \leqslant r\}| \leqslant Ce^{Cr}. \tag{6.4}$$

Thanks to (6.4), the Ruelle dynamical zeta function is well defined for  $\sigma \in \mathbf{C}$  with Re  $(\sigma) \gg 1$ . Recall that for a prime closed orbit  $\gamma$ ,  $\Delta(\gamma) \in \{\pm 1\}$  is defined in (4.9). Proceeding as in the proof of Proposition 4.5, for  $\sigma \in \mathbf{C}$  with Re  $(\sigma) \gg 1$ , we have

$$R_{\phi,\rho}(\sigma) = \left(\prod_{\gamma: \text{prime}} \det\left(1 - \Delta(\gamma)\rho(\gamma)e^{-\sigma\ell_{\gamma}}\right)\right)^{(-1)^{\text{rk}[E^u]+1}}, \tag{6.5}$$

where rk  $[E^u]$  denotes the rank of the  $C^0$ -vector bundle  $E^u$ . We note the similarity between (0.2) and (6.5).

**Example 6.2.** If  $(Z, g^{TZ})$  is a negatively curved manifold, then the geodesic flow on the unit tangent bundle M = SZ is Anosov [A67]. By (6.5), for  $\sigma \in \mathbf{C}$  with  $\operatorname{Re}(\sigma) \gg 1$ ,

$$R_{\phi,\rho}(\sigma) = \left( \prod_{\gamma: \text{prime}} \det \left( 1 - \Delta(\gamma)\rho(\gamma)e^{-\sigma\ell_{\gamma}} \right) \right)^{(-1)^{\dim Z}}.$$
 (6.6)

In addition, if Z is orientable, for all prime closed orbits  $\gamma$ , we have  $\Delta(\gamma) = 1$  (see [GiLiPo13, Lemma B.1]).

### 6.2. The meromorphic extension

The proofs of the meromorphic extension of the Ruelle dynamical zeta function for the Anosov flow given in [GiLiPo13] and [DyZ16]<sup>13</sup> are based on a spectral interpretation of  $R_{\phi,\rho}$ .

Let us begin with establishing a relation between  $R_{\phi,\rho}$  and the Lie derivation  $L_V$  along V acting on  $\Omega^{\cdot}(M,F)$ . For t>0, write

$$e^{tL_V}: u \in \Omega^{\cdot}(M, F) \to \phi_t^* u \in \Omega^{\cdot}(M, F). \tag{6.7}$$

<sup>&</sup>lt;sup>13</sup>Unfortunately, there is a sign conflict between the convention used in these two papers and Fried's paper [F87]. In [GiLiPo13, (2.2)] and [DyZ16, (1.1),(B.1)], the sign in the Fuller index (2.18) is defined using  $D\phi_{-\ell\gamma}$ . As we adopt Fried's convention, the statements in this section are slightly different with [GiLiPo13, DyZ16].

The Schwartz kernel  $e^{tL_V}(x,y)$  of  $e^{tL_V}$  is a current on  $\mathbf{R}_+^* \times M \times M$  with coefficients in

$$\mathbf{C} \boxtimes (\Lambda^{\cdot}(T^*M) \otimes_{\mathbf{R}} F) \boxtimes \Big( (\Lambda^{\cdot}(T^*M) \otimes_{\mathbf{R}} F)^* \otimes_{\mathbf{R}} |\det(T^*M)| \Big), \tag{6.8}$$

where  $|\det(T^*M)| = \Lambda^{\dim M}(T^*M) \otimes o(TM)$  is the density bundle on M. By (6.1), its wave front set is disjoint from the conormal bundle of the submanifold  $\{(t, x, x) \in \mathbf{R}_+^* \times M \times M\} \subset \mathbf{R}_+^* \times M \times M$ . So, the restriction on the diagonal  $e^{tL_V}(x, x)$  is a well-defined current on  $\mathbf{R}_+^* \times M$  with coefficients in

$$\mathbf{C} \boxtimes \Big( \operatorname{End} \left( F \otimes_{\mathbf{R}} \Lambda^{\cdot}(T^*M) \right) \otimes_{\mathbf{R}} |\det(T^*M)| \Big).$$
 (6.9)

**Definition 6.3.** The flat trace of  $e^{tL_V}$  is a distribution on  $\mathbf{R}_{+}^*$  defined by

$$\operatorname{Tr}^{\flat}\left[e^{tL_{V}}\right] = \int_{M} \operatorname{Tr}\left[e^{tL_{V}}(x,x)\right]. \tag{6.10}$$

Note that the flat trace is not a classical trace in the sense of the trace of a trace class operator. However, we can still show that the flat trace of the commutator of  $e^{tL_V}$  with a differential operator vanishes.

Let us give an explicit formula for  $\operatorname{Tr}^{\flat}\left[e^{tL_{V}}\right]$ , which is known as the Atiyah–Bott–Guillemin trace formula [Gui77]. Let  $\gamma\in\overline{\wp}(\phi)$  be a closed orbit. For  $x\in\gamma$ , we have a morphism

$$D\phi_{\ell_{\gamma}}(x): T_x M \to T_x M.$$
 (6.11)

Up to conjugation it does not depend on the choice of  $x \in \gamma$ , and is denoted by  $D\phi_{\ell_{\gamma}}|_{T_{\gamma}M}$ . It acts naturally on any tensor of  $T_{\gamma}M$ . Acting on  $T_{\gamma}M/\mathbf{R}V$ , it is just the linearized Poincaré return map.

For  $0 \leq q \leq \dim M$ , denote by  $L_{V,q}$  the restriction of  $L_V$  on  $\Omega^q(M,F)$ . Let  $N^{\Lambda^*(T^*M)}$  be the number operator on  $\Lambda^*(T^*M)$ , which sends  $s \in \Omega^q(M,F)$  to  $qs \in \Omega^q(M,F)$ .

**Proposition 6.4.** For  $0 \le q \le \dim M$ , the following identity holds,

$$\operatorname{Tr}^{\flat}\left[e^{tL_{V,q}}\right] = \sum_{\gamma \in \overline{\wp}(\phi.)} \frac{\operatorname{Tr}^{\Lambda^{q}(T_{\gamma}^{*}M)}\left[\left(D\phi_{\ell_{\gamma}}\right)^{\operatorname{tr}}|_{\Lambda^{q}(T_{\gamma}^{*}M)}\right]}{|\det(1 - D\phi_{\ell_{\gamma}})|_{T_{\gamma}M/\mathbf{R}V}|} \operatorname{Tr}\left[\rho(\gamma)\right] \frac{\ell_{\gamma}}{m_{\gamma}} \delta_{\ell_{\gamma}}(t). \quad (6.12)$$

In particular, we have

$$\operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda^{\cdot} (T^{*}M)} e^{tL_{V}} \right] = -\sum_{\gamma \in \overline{\wp}(\phi.)} \operatorname{sgn} \left( \det(1 - D\phi_{\ell_{\gamma}}) |_{T_{\gamma}M/\mathbf{R}V} \right) \operatorname{Tr} \left[ \rho(\gamma) \right] \frac{\ell_{\gamma}}{m_{\gamma}} \delta_{\ell_{\gamma}}(t).$$

$$(6.13)$$

*Proof.* The Schwartz kernel of  $e^{tL_{V,q}}$  is given by

$$e^{tL_{V,q}}(x,y) = \left\{ \tau_0^t \otimes (D\phi_t)^{\text{tr}} |_{\Lambda^q \left(T_{\phi_t(x)}^* M\right)} \right\} \cdot \delta_{\phi_t(x)}(y), \tag{6.14}$$

where  $\tau_0^t \in \text{Hom}(F_{\phi_t(x)}, F_x)$  is the parallel transport with respect to  $\nabla^F$  along the curve  $(\phi_s(x))_{0 \leqslant s \leqslant t}$  from t to 0. So the restriction of the distribution to the diagonal  $e^{tL_V}(x,x)$  is supported on the periodic set  $\wp(\phi) \subset \mathbf{R}_+^* \times M$  and is given by 14

$$\operatorname{Tr}\left[e^{tL_{V,q}}(x,x)\right] = \sum_{\gamma \in \overline{\wp}(\phi_{\cdot})} \operatorname{Tr}\left[\rho(\gamma)\right] \frac{\operatorname{Tr}^{\Lambda^{q}\left(T_{\gamma}^{*}M\right)}\left[\left(D\phi_{\ell_{\gamma}}\right)^{\operatorname{tr}}|_{\Lambda^{q}\left(T_{\gamma}^{*}M\right)}\right]}{|\det(1 - D\phi_{\ell_{\gamma}})|_{T_{\gamma}M/\mathbf{R}V}|} \delta_{\ell_{\gamma}}(t) \otimes \delta_{\gamma^{\sharp}}(x),$$
(6.15)

where  $\delta_{\gamma^{\sharp}}(x)$  is the current of integration on the prime closed orbit  $\gamma^{\sharp}$  associated to  $\gamma$  defined by

$$s \in C_c^{\infty}(M) \to \int_0^{\ell_{\gamma}/m_{\gamma}} s(\phi_t(x)) dt,$$
 (6.16)

where x is any point on  $\gamma$ . By (6.15), we get (6.12).

By (2.21), (6.4), and (6.13), for Re  $(\sigma) \gg 1$ , we have

$$\log R_{\phi,\rho}(\sigma) = -\int_0^\infty \operatorname{Tr}_s^{\flat} \left[ N^{\Lambda^*(T^*M)} \exp\left(-t\left(\sigma - L_V\right)\right) \right] \frac{dt}{t}. \tag{6.17}$$

So formally,  $R_{\phi,\rho}(\sigma)$  is a certain flat regularized determinant

$$\prod_{q=1}^{\dim M} \det^{\flat} \left( \sigma - L_{V,q} \right)^{(-1)^{q} q}. \tag{6.18}$$

Note the similarity between (1.8) and (6.18).

For Re  $(\sigma) \gg 1$ , we write

$$\frac{\partial}{\partial \sigma} \log R_{\phi,\rho}(\sigma) = \int_{0}^{\infty} \operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda \cdot (T^{*}M)} \exp\left(-t\left(\sigma - L_{V}\right)\right) \right] dt$$

$$= \int_{s}^{\infty} \operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda \cdot (T^{*}M)} \exp\left(-t\left(\sigma - L_{V}\right)\right) \right] dt,$$
(6.19)

where  $\delta > 0$  is some positive number smaller than the minimum of the length spectrum. In the second identity of (6.19), we use the fact that the support of the distribution  $\operatorname{Tr}_{s}^{\flat}\left[N^{\Lambda^{*}(T^{*}M)}\exp\left(-t\left(\sigma-L_{V}\right)\right)\right]$  is away from 0.

To show the meromorphic extension of  $R_{\phi,\rho}$ , it is enough to show that  $\frac{\partial}{\partial \sigma} \log R_{\phi,\rho}$  has a meromorphic extension to **C** with simple poles and integer residues. By (6.19), we write formally

$$\frac{\partial}{\partial \sigma} \log R_{\phi,\rho}(\sigma) = \operatorname{Tr}_{\mathrm{s}}^{\flat} \left[ N^{\Lambda^{\bullet}(T^{*}M)} e^{-\delta(\sigma - L_{V})} \left( \sigma - L_{V} \right)^{-1} \right]. \tag{6.20}$$

An important step is to give a proper sense of the operators on the right-hand side of (6.20) and to show its flat trace exists and has a meromorphic extension. We refer the reader to [DyZ16] for more details. Here we just state a weak version

 $<sup>^{14}</sup>$ This can be obtained by proceeding as in the proof of Lefschetz fixed point formula. We refer the reader to  $[\mathrm{DyZ16}, \mathrm{Appendix} \ \mathrm{B}]$  for more details.

of [DyZ16, Propositions 3.1–3.3], and explain the reason for which we need to introduce the small  $\delta > 0$ .

Let

$$T^*M = (\mathbf{R}V)^* \oplus E_u^* \oplus E_s^* \tag{6.21}$$

be the dual of the splitting (6.1). Let  $\Delta \subset T^*(M \times M)$  be the diagonal of  $T^*(M \times M)$ . Set

$$\Omega^{-} = \left\{ \left( \phi_t(x), \left( (D\phi_t)_x^{\mathrm{tr}} \right)^{-1} \cdot \xi, x, \xi \right) \in T^*(M \times M) : \langle V(x), \xi \rangle = 0, t \leqslant 0 \right\}, \tag{6.22}$$

where ·tr denotes the transpose of a matrix. Denote by WF' the wave front set of an operator (see [DyZ16, Appendix C.2]).

**Theorem 6.5.** The operator

$$(\sigma - L_V)^{-1} : \Omega^{\cdot}(M, F) \to \mathcal{D}'(M, \Lambda^{\cdot}(T^*M) \otimes_{\mathbf{R}} F)$$
(6.23)

defines a meromorphic family on C. If it is holomorphic at  $\sigma_0$ , then

$$WF'(\sigma_0 - L_V)^{-1} \subset \Delta \cup \Omega^- \cup (E_s^* \times E_u^*). \tag{6.24}$$

From (6.24), we see that  $(\sigma_0 - L_V)^{-1}$  does not necessarily have a well-defined flat trace. But  $e^{-\delta(\sigma - L_V)} (\sigma - L_V)^{-1}$  does.

# 6.3. The $R_{\phi,\rho}(0)$ as a topological invariant

The poles of the meromorphic family in (6.23) are called Ruelle–Pollicott resonances. The set of Ruelle–Pollicott resonances is denoted by  $\operatorname{Res}_{\rho}(V)$ . If  $0 \notin \operatorname{Res}_{\rho}(V)$ , then  $R_{\phi,\rho}(\sigma)$  is regular at  $\sigma = 0.15$ 

Set

$$\mathcal{V}_{\rho}(M) = \{ V \in C^{\infty}(M, TM) : V \text{ is Anosov such that } 0 \notin \operatorname{Res}_{\rho} V \}. \tag{6.25}$$

Thanks to the stability of the Anosov flow [A67] and of its resonance [ButLi07, ButLi13],  $\mathcal{V}_{\rho}(M)$  forms an open subset in  $C^{\infty}(M,TM)$ . The following theorem [DaGuRiSh20] tells us the value at zero of the Ruelle dynamical zeta function does not depend on a small perturbation of the flow.

For  $V \in \mathscr{V}_{\rho}(M)$ , denote by  $\phi^{V}$  the corresponding flow.

**Theorem 6.6.** For any flat vector bundle F, the map

$$V \in \mathscr{V}_{\rho}(M) \to R_{\phi^{V},\rho}(0) \in \mathbf{C}^*$$
 (6.26)

is locally constant.

<sup>&</sup>lt;sup>15</sup>Due to the cancellation from the supertrace, the converse is not correct.

*Proof.* Let  $(V_b)_{b\in\mathbf{R}}$  be a smooth family of vector fields in  $\mathscr{V}_{\rho}(M)$ . Let  $R_{b,\rho}$  be the corresponding family of the Ruelle dynamical zeta functions. Take a smooth family  $\alpha_b \in \Omega^1(M)$  such that  $\alpha_b(V_b) = 1$ . Write  $\dot{V}_b = \frac{\partial}{\partial b} V_b$ .

We claim that for  $\sigma \in \mathbf{C}$  with  $\operatorname{Re}(\sigma) \gg 1$ , we have

$$\frac{\partial}{\partial b} \log R_{b,\rho}(\sigma) = -\sigma \int_{\delta}^{\infty} \operatorname{Tr}_{s}^{\flat} \left[ \alpha_{b} i_{\dot{V}_{b}} \exp \left( -t \left( \sigma - L_{V_{b}} \right) \right) \right] dt. \tag{6.27}$$

In [DaGuRiSh20], the proof of (6.27) is obtained by variation of the periods of closed orbits. Here, we give a proof via supersymmetry (cf. [RS71, Theorem 2.1]). We argue formally. The argument can be made rigorous easily. By (6.17), for  $\text{Re}(\sigma) \gg 1$ , we have

$$\frac{\partial}{\partial b} \log R_{b,\rho}(\sigma) = -\int_{\delta}^{\infty} \operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda \cdot (T^{*}M)} L_{\dot{V}_{b}} \exp\left(-t \left(\sigma - L_{V_{b}}\right)\right) \right] dt. \tag{6.28}$$

Using the Cartan identity<sup>16</sup>  $L_{\dot{V}_b} = [d, i_{\dot{V}_b}]$ , the fact that d commutes with  $L_{V_b}$ , and fact that the supertrace vanishes on the supercommutator, we have identities of distributions on  $\mathbf{R}_{+}^*$ ,

$$\operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda^{\bullet}(T^{*}M)} L_{\dot{V}_{b}} \exp\left(tL_{V_{b}}\right) \right] = \operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda^{\bullet}(T^{*}M)} \left[d, i_{\dot{V}_{b}}\right] \exp\left(tL_{V_{b}}\right) \right]$$

$$= \operatorname{Tr}_{s}^{\flat} \left[ \left[ N^{\Lambda^{\bullet}(T^{*}M)}, d \right] i_{\dot{V}_{b}} \exp\left(tL_{V_{b}}\right) \right]$$

$$= \operatorname{Tr}_{s}^{\flat} \left[ di_{\dot{V}_{c}} \exp\left(tL_{V_{b}}\right) \right].$$

$$(6.29)$$

Since  $\alpha_b(V_b) = 1$ , we have  $i_{\dot{V}_b} = [i_{V_b}, \alpha_b i_{\dot{V}_b}]$ . By (6.29), and proceeding as before we have

$$\operatorname{Tr}_{s}^{\flat} \left[ N^{\Lambda^{\bullet}(T^{*}M)} L_{\dot{V}_{b}} \exp(tL_{V_{b}}) \right] = \operatorname{Tr}_{s}^{\flat} \left[ d \left[ i_{V_{b}}, \alpha_{b} i_{\dot{V}_{b}} \right] \exp(tL_{V_{b}}) \right]$$

$$= \operatorname{Tr}_{s}^{\flat} \left[ \left[ d, i_{V_{b}} \right] \alpha_{b} i_{\dot{V}_{b}} \exp(tL_{V_{b}}) \right]$$

$$= \operatorname{Tr}_{s}^{\flat} \left[ L_{V_{b}} \alpha_{b} i_{\dot{V}_{b}} \exp(tL_{V_{b}}) \right]$$

$$= \frac{\partial}{\partial t} \operatorname{Tr}_{s}^{\flat} \left[ \alpha_{b} i_{\dot{V}_{b}} \exp(tL_{V_{b}}) \right].$$

$$(6.30)$$

By (6.28) and (6.30), we get (6.27).

By (6.27), using the method as in [DyZ16], we can show that for  $\sigma \in \mathbf{C}$ , Re  $(\sigma) \gg 1$ ,

$$\frac{\partial}{\partial b} \log R_{b,\rho}(\sigma) = -\sigma \operatorname{Tr}_{s}^{\flat} \left[ \alpha_{b} i_{\dot{V}_{b}} e^{-\delta(\sigma - L_{V_{b}})} \left( \sigma - L_{V_{b}} \right)^{-1} \right]. \tag{6.31}$$

Note that thanks to (6.24), the above flat trace is well defined. Moreover, the function  $\sigma \to \operatorname{Tr}_{\mathbf{s}}^{\flat} \left[ \alpha_b i_{\dot{V}_b} e^{-\delta(\sigma - L_{V_b})} \left( \sigma - L_{V_b} \right)^{-1} \right]$  has a meromorphic extension to  $\mathbf{C}$ , and is regular at  $\sigma = 0$  since  $V_b \in \mathscr{V}_{\rho}(M)$ .

<sup>&</sup>lt;sup>16</sup>Here  $[a,b] = ab - (-1)^{\deg a \deg b} ba$  denotes the supercommutator of a and b (see [BeGeVe04, Section 1.3]).

To complete our proof, it remains to show that for a fixed  $\sigma$  near 0, the function  $b \to R_{b,\rho}(\sigma)$  is  $C^1$  and (6.31) holds near 0. This is somehow technical and we refer the reader to [DaGuRiSh20] for a detailed proof.

### 6.4. Anosov flow on 3-manifold

Let us restrict ourself to an orientable 3-manifold, where we have a partial solution for the Fried conjecture.

The following proposition [DaGuRiSh20, Proposition 7.3, Lemma 7.4] gives a characterization of the acyclicity of a unitarily flat vector bundle via resonance. Its proof uses [DyZ17, Lemma 2.3] in an essential way.

**Proposition 6.7.** Let F be a unitarily flat vector bundle on a closed orientable 3-manifold. For any volume preserving Anosov flow, F is acyclic if and only if 0 is not a resonance.

Recall the following theorem due to Sánchez-Morgado [SM96a], whose proof is based on the Markov partition [Rat69] and Rugh's technique [Rug96].

**Theorem 6.8.** Let F be an acyclic unitarily flat vector bundle with holonomy  $\rho$  on a closed orientable analytic 3-manifold M. If  $\phi$ , is a transitive analytic Anosov flow, and if there is a prime closed orbit  $\gamma$  such that 1 and  $\Delta(\gamma)$  are not eigenvalues of  $\rho(\gamma)$ , then  $R_{\phi,\rho}(\sigma)$  is regular at 0 and

$$|R_{\phi,\rho}(0)| = T_F(M).$$
 (6.32)

Note that any smooth manifold has a unique compatible analytic structure, and that any volume preserving Anosov flow is transitive. Since we can always approximate a smooth Anosov flow by an analytic one, and since we can approximate a flat vector bundle by the one with specified holonomy condition in the above theorem provided  $H^1(M) \neq 0$ , using Theorem 6.6, in [DaGuRiSh20, Section 7.2], we deduce the following Theorem [DaGuRiSh20, Theorem 1].

**Theorem 6.9.** Let F be an acyclic unitarily flat vector bundle with holonomy  $\rho$  on a closed orientable 3-manifold with  $H^1(M) \neq 0$ . For any flow  $\phi$ , which is a volume preserving Anosov flow or a flow nearby  $^{17}$ , we have

$$|R_{\phi,\rho}(0)| = T_F(M).$$
 (6.33)

Let us return to the case of the geodesic flow on the unit tangent bundle M = SZ of a negatively curved orientable surface  $(Z, g^{TZ})$ . Recall that  $a_0 \in \pi_1(M)$  is defined after (5.8). By Corollary 5.4 and Theorem 6.9, we get:

**Corollary 6.10.** Let F be an acyclic unitarily flat vector bundle on the unit tangent bundle of a negatively curved orientable surface  $(Z, q^{TZ})$ . Then,

$$|R_{\phi,\rho}(0)| = T_F(M) = \left| \det (1 - \rho(a_0)) \right|^{-\chi(Z)}.$$
 (6.34)

The above corollary can be considered as a complementary of Dyatlov–Zworski's result [DyZ17], where  $\rho$  is assumed to be trivial.

<sup>&</sup>lt;sup>17</sup>It is still an Anosov flow by the stability of Anosov flows [A67].

**Theorem 6.11.** Assume that  $(Z, g^{TZ})$  is a negatively curved orientable surface. There is  $C \in \mathbb{R}^*$  such that as  $\sigma \to 0$ , we have

$$R_{\phi,\text{trivial}}(\sigma) = C\sigma^{-\chi(Z)}(1 + \mathcal{O}(\sigma)).$$
 (6.35)

The above two results are generalizations of Fried's results [F86b, Corollaries 1 and 2] for hyperbolic surfaces.

## References

- [A67] D.V. Anosov, Geodesic flows on closed Riemannian manifolds of negative curvature, Trudy Mat. Inst. Steklov. 90 (1967), 209. MR 0224110
- [ArMaz65] M. Artin and B. Mazur, On periodic points, Ann. of Math. (2) 81 (1965), 82–99. MR 0176482
- [Ba18] V. Baladi, Dynamical zeta functions and dynamical determinants for hyperbolic maps, Ergebnisse der Mathematik und ihrer Grenzgebiete. 3. Folge, vol. 68, Springer International Publishing, 2018.
- [BaGrSch85] W. Ballmann, M. Gromov, and V. Schroeder, Manifolds of nonpositive curvature, Progress in Mathematics, vol. 61, Birkhäuser Boston, Inc., Boston, MA, 1985. MR 823981
- [BeGeVe04] N. Berline, E. Getzler, and M. Vergne, Heat kernels and Dirac operators, Grundlehren Text Editions, Springer-Verlag, Berlin, 2004, Corrected reprint of the 1992 original. MR 2273508 (2007m:58033)
- [B05] J.-M. Bismut, The hypoelliptic Laplacian on the cotangent bundle, J. Amer. Math. Soc. 18 (2005), no. 2, 379–476 (electronic). MR 2137981 (2006f:35036)
- [B11] J.-M. Bismut, Hypoelliptic Laplacian and orbital integrals, Annals of Mathematics Studies, vol. 177, Princeton University Press, Princeton, NJ, 2011. MR 2828080
- [B19] J.-M. Bismut, Eta invariants and the hypoelliptic Laplacian, J. Eur. Math. Soc. 21 (2019), 2355–2515.
- [BG01] J.-M. Bismut and S. Goette, Families torsion and Morse functions, Astérisque (2001), no. 275, x+293. MR 1867006 (2002h:58059)
- [BG04] J.-M. Bismut and S. Goette, Equivariant de Rham torsions, Ann. of Math.
   (2) 159 (2004), no. 1, 53–216. MR 2051391 (2005f:58059)
- [BL095] J.-M. Bismut and J. Lott, Flat vector bundles, direct images and higher real analytic torsion, J. Amer. Math. Soc. 8 (1995), no. 2, 291–363. MR 1303026 (96g:58202)
- [BZ92] J.-M. Bismut and W. Zhang, An extension of a theorem by Cheeger and Müller, Astérisque (1992), no. 205, 235, With an appendix by François Laudenbach. MR 1185803 (93j:58138)
- [BZ94] J.-M. Bismut and W. Zhang, Milnor and Ray-Singer metrics on the equivariant determinant of a flat vector bundle, Geom. Funct. Anal. 4 (1994), no. 2, 136–212. MR 1262703 (96f:58179)
- [BrMa13] J. Brüning and X. Ma, On the gluing formula for the analytic torsion, Math. Z. 273 (2013), no. 3-4, 1085-1117. MR 3030691

- [Bo65] R. Bott, The index theorem for homogeneous differential operators, Differential and Combinatorial Topology (A Symposium in Honor of Marston Morse), Princeton Univ. Press, Princeton, N.J., 1965, pp. 167–186. MR 0182022 (31 #6246)
- [BoTu82] R. Bott and L.W. Tu, Differential forms in algebraic topology, Graduate Texts in Mathematics, vol. 82, Springer-Verlag, New York-Berlin, 1982. MR 658304
- [BorW00] A. Borel and N. Wallach, Continuous cohomology, discrete subgroups, and representations of reductive groups, second ed., Mathematical Surveys and Monographs, vol. 67, American Mathematical Society, Providence, RI, 2000. MR 1721403 (2000j:22015)
- [BuO95] U. Bunke and M. Olbrich, Selberg zeta and theta functions, Mathematical Research, vol. 83, Akademie-Verlag, Berlin, 1995, A differential operator approach. MR 1353651 (97c:11088)
- [ButLi07] O. Butterley and C. Liverani, Smooth Anosov flows: correlation spectra and stability, J. Mod. Dyn. 1 (2007), no. 2, 301–322. MR 2285731
- [ButLi13] O. Butterley and C. Liverani, Robustly invariant sets in fiber contracting bundle flows, J. Mod. Dyn. 7 (2013), no. 2, 255–267. MR 3106713
- [BWSh20] Y. Borns-Weil and S. Shen, Dynamical zeta functions in the nonorientable case, arXiv:2007.08043 (2020).
- [C79] J. Cheeger, Analytic torsion and the heat equation, Ann. of Math. (2) 109 (1979), no. 2, 259–322. MR 528965 (80j:58065a)
- [DMe12] X. Dai and R.B. Melrose, Adiabatic limit, heat kernel and analytic torsion, Metric and differential geometry, Progr. Math., vol. 297, Birkhäuser/Springer, Basel, 2012, pp. 233–298. MR 3220445
- [DaGuRiSh20] N. V. Dang, C. Guillarmou, G. Rivière, and S. Shen, The Fried conjecture in small dimensions, Invent. Math. 220 (2020), no. 2, 525–579. MR 4081137
- [DaRi19] N. V. Dang and G. Rivière, Spectral analysis of Morse-Smale gradient flows, Ann. Sci. Éc. Norm. Supér. (4) 52 (2019), no. 6, 1403–1458. MR 4061023
- [DaRi20a] N. V. V. Dang and G. Rivière, Spectral analysis of Morse-Smale flows, I: construction of the anisotropic spaces, J. Inst. Math. Jussieu 19 (2020), no. 5, 1409–1465. MR 4138948
- [DaRi20b] N.V. Dang and G. Rivière, Spectral analysis of Morse-Smale flows, II: Resonances and resonant states, Amer. J. Math. 142 (2020), no. 2, 547–593. MR 4084163
- [DaRi20c] N.V. Dang and G. Rivière, Topology of Pollicott-Ruelle resonant states, to appear in Ann. Sc. Norm. Super. Pisa Cl. Sci, doi: 10.2422/2036-2145.201804\_010
- [DaRi21] N.V. Dang and G. Rivière, Pollicott-Ruelle spectrum and Witten L aplacians, to appear in J. Eur. Math. Soc. doi: 10.4171/JEMS/1044
- [dR50] G. de Rham, Complexes à automorphismes et homéomorphie différentiable, Ann. Inst. Fourier Grenoble 2 (1950), 51–67 (1951). MR 0043468 (13,268c)
- [DuKV79] J.J. Duistermaat, J.A.C. Kolk, and V.S. Varadarajan, Spectra of compact locally symmetric manifolds of negative curvature, Invent. Math. 52 (1979), no. 1, 27–93. MR 532745 (82a:58050a)

- [DyGu16] S. Dyatlov and C. Guillarmou, Pollicott-Ruelle resonances for open systems, Ann. Henri Poincaré 17 (2016), no. 11, 3089–3146. MR 3556517
- [DyGu18] S. Dyatlov and C. Guillarmou, Afterword: dynamical zeta functions for Axiom A flows, Bull. Amer. Math. Soc. (N.S.) 55 (2018), no. 3, 337–342. MR 3803156
- [DyZ16] S. Dyatlov and M. Zworski, Dynamical zeta functions for Anosov flows via microlocal analysis, Ann. Sci. Éc. Norm. Supér. (4) 49 (2016), no. 3, 543–577. MR 3503826
- [DyZ17] S. Dyatlov and M. Zworski, Ruelle zeta function at zero for surfaces, Invent. Math. 210 (2017), no. 1, 211–229. MR 3698342
- [FaT17] F. Faure and M. Tsujii, The semiclassical zeta function for geodesic flows on negatively curved manifolds, Invent. Math. 208 (2017), no. 3, 851–998. MR 3648976
- [Fra82] J.M. Franks, Homology and dynamical systems, CBMS Regional Conference Series in Mathematics, vol. 49, Published for the Conference Board of the Mathematical Sciences, Washington, D.C.; by the American Mathematical Society, Providence, R. I., 1982. MR 669378
- [Fr35] W. Franz, Über die Torsion einer Überdeckung, J. Reine Angew. Math. 173 (1935), 245–254 (German).
- [F86a] D. Fried, Analytic torsion and closed geodesics on hyperbolic manifolds, Invent. Math. 84 (1986), no. 3, 523-540. MR 837526 (87g:58118)
- [F86b] D. Fried, Fuchsian groups and Reidemeister torsion, The Selberg trace formula and related topics (Brunswick, Maine, 1984), Contemp. Math., vol. 53, Amer. Math. Soc., Providence, RI, 1986, pp. 141–163. MR 853556 (88e:58098)
- [F86c] D. Fried, The zeta functions of Ruelle and Selberg. I, Ann. Sci. École Norm. Sup. (4) 19 (1986), no. 4, 491–517. MR 875085
- [F87] D. Fried, Lefschetz formulas for flows, The Lefschetz centennial conference, Part III (Mexico City, 1984), Contemp. Math., vol. 58, Amer. Math. Soc., Providence, RI, 1987, pp. 19–69. MR 893856 (88k:58138)
- [F88] D. Fried, Torsion and closed geodesics on complex hyperbolic manifolds, Invent. Math. 91 (1988), no. 1, 31–51. MR 918235
- [F95] D. Fried, Meromorphic zeta functions for analytic flows, Comm. Math. Phys. 174 (1995), no. 1, 161–190. MR 1372805
- [Fu67] F.B. Fuller, An index of fixed point type for periodic orbits, Amer. J. Math. 89 (1967), 133–148. MR 0209600
- [GeGraPS69] I.M. Gel'fand, M.I. Graev, and I.I. Pyatetskii-Shapiro, Representation theory and automorphic functions, Translated from the Russian by K.A. Hirsch, W.B. Saunders Co., Philadelphia, Pa.-London-Toronto, Ont., 1969. MR 0233772 (38 #2093)
- [GiLiPo13] P. Giulietti, C. Liverani, and M. Pollicott, Anosov flows and dynamical zeta functions, Ann. of Math. (2) 178 (2013), no. 2, 687–773. MR 3071508
- [Gui77] V. Guillemin, Lectures on spectral theory of elliptic operators, Duke Math. J. 44 (1977), no. 3, 485–517. MR 0448452
- [H66] F. Hirzebruch, Topological methods in algebraic geometry, Third enlarged edition. New appendix and translation from the second German edition by R.L.E.

- Schwarzenberger, with an additional section by A. Borel. Die Grundlehren der Mathematischen Wissenschaften, Band 131, Springer-Verlag New York, Inc., New York, 1966. MR 0202713
- [K86] A.W. Knapp, Representation theory of semisimple groups, Princeton Mathematical Series, vol. 36, Princeton University Press, Princeton, NJ, 1986, An overview based on examples. MR 855239 (87j:22022)
- [K02] A.W. Knapp, Lie groups beyond an introduction, second ed., Progress in Mathematics, vol. 140, Birkhäuser Boston, Inc., Boston, MA, 2002. MR 1920389 (2003c:22001)
- [LoRo91] J. Lott and M. Rothenberg, Analytic torsion for group actions, J. Differential Geom. 34 (1991), no. 2, 431–481. MR 1131439
- [Lü93] W. Lück, Analytic and topological torsion for manifolds with boundary and symmetry, J. Differential Geom. 37 (1993), no. 2, 263–322. MR 1205447
- [LüScTh98] W. Lück, T. Schick, and T. Thielmann, Torsion and fibrations, J. Reine Angew. Math. 498 (1998), 1–33. MR 1629917
- [Ma02] X. Ma, Functoriality of real analytic torsion forms, Israel J. Math. 131 (2002), 1–50. MR 1942300
- [Ma17] X. Ma, Geometric hypoelliptic Laplacian and orbital integral, after Bismut, Lebeau and Shen, Séminaire Bourbaki, No. 1130, March 11 2017. available at http://www.bourbaki.ens.fr/TEXTES/1130.pdf. Video available at https://www.youtube.com/watch?v=dCDwN-HqcJw.
- [MaMari15] X. Ma and G. Marinescu, Exponential estimate for the asymptotics of Bergman kernels, Math. Ann. 362 (2015), no. 3-4, 1327–1347. MR 3368102
- [Mar04] G.A. Margulis, On some aspects of the theory of Anosov systems, Springer Monographs in Mathematics, Springer-Verlag, Berlin, 2004, With a survey by Richard Sharp: Periodic orbits of hyperbolic flows, Translated from the Russian by Valentina Vladimirovna Szulikowska. MR 2035655
- [Mi68a] J. Milnor, Infinite cyclic coverings, Conference on the Topology of Manifolds (Michigan State Univ., E. Lansing, Mich., 1967), Prindle, Weber & Schmidt, Boston, Mass., 1968, pp. 115–133. MR 0242163 (39 #3497)
- [Mi68b] J. Milnor, A note on curvature and fundamental group, J. Differential Geometry 2 (1968), 1–7. MR 0232311
- [Mil78] J.J. Millson, Closed geodesics and the  $\eta$ -invariant, Ann. of Math. (2) 108 (1978), no. 1, 1–39. MR 0501204
- [MoSt89] H. Moscovici and R.J. Stanton, Eta invariants of Dirac operators on locally symmetric manifolds, Invent. Math. 95 (1989), no. 3, 629–666. MR 979370
- [MoSt91] H. Moscovici and R.J. Stanton, R-torsion and zeta functions for locally symmetric manifolds, Invent. Math. 105 (1991), no. 1, 185–216. MR 1109626 (92i:58199)
- [MoSt18] H. Moscovici and R.J. Stanton, Holomorphic torsion with coefficients and geometric zeta functions for certain hermitian locally symmetric manifolds, appendix by J. Frahm, arXiv:1802.08886(2018)
- [M78] W. Müller, Analytic torsion and R-torsion of Riemannian manifolds, Adv. in Math. 28 (1978), no. 3, 233–305. MR 498252 (80j:58065b)

- [M93] W. Müller, Analytic torsion and R-torsion for unimodular representations, J. Amer. Math. Soc. 6 (1993), no. 3, 721–753. MR 1189689 (93m:58119)
- [M12] W. Müller, The asymptotics of the Ray-Singer analytic torsion of hyperbolic 3-manifolds, Metric and differential geometry, Progr. Math., vol. 297, Birkhäuser/Springer, Basel, 2012, pp. 317–352. MR 3220447
- [M20] W. Müller, On Fried's conjecture for compact hyperbolic manifolds, arXiv: 2005.01450 (2020).
- [Pa09] J. Park, Analytic torsion and Ruelle zeta functions for hyperbolic manifolds with cusps, J. Funct. Anal. 257 (2009), no. 6, 1713–1758. MR 2540990
- [PalMe82] J. Palis, Jr. and W. de Melo, Geometric theory of dynamical systems, Springer-Verlag, New York-Berlin, 1982, An introduction, Translated from the Portuguese by A.K. Manning. MR 669541
- [Rat69] M.E. Ratner, Markov decomposition for an Y-flow on a three-dimensional manifold, Mat. Zametki 6 (1969), 693–704. MR 0260977
- [RS71] D.B. Ray and I.M. Singer, R-torsion and the Laplacian on Riemannian manifolds, Advances in Math. 7 (1971), 145–210. MR 0295381 (45 #4447)
- [Re35] K. Reidemeister, Homotopieringe und Linsenräume, Abh. Math. Sem. Univ. Hamburg 11 (1935), no. 1, 102–109. MR 3069647
- [Ru76a] D. Ruelle, Generalized zeta-functions for Axiom A basic sets, Bull. Amer. Math. Soc. 82 (1976), no. 1, 153–156. MR 0400311
- [Ru76b] D. Ruelle, Zeta-functions for expanding maps and Anosov flows, Invent. Math. 34 (1976), no. 3, 231–242. MR 0420720
- [Rug96] H.H. Rugh, Generalized Fredholm determinants and Selberg zeta functions for Axiom A dynamical systems, Ergodic Theory Dynam. Systems 16 (1996), no. 4, 805–819. MR 1406435
- [SM93] H. Sánchez-Morgado, Lefschetz formulae for Anosov flows on 3-manifolds, Ergodic Theory Dynam. Systems 13 (1993), no. 2, 335–347. MR 1235476
- [SM96a] H. Sánchez-Morgado, R-torsion and zeta functions for analytic Anosov flows on 3-manifolds, Trans. Amer. Math. Soc. 348 (1996), no. 3, 963–973. MR 1348868
- [SM96b] H. Sánchez-Morgado, Reidemeister torsion and Morse-Smale flows, Ergodic Theory Dynam. Systems 16 (1996), no. 2, 405-414. MR 1389631
- [SR99] S.A. Salamanca-Riba, On the unitary dual of real reductive Lie groups and the  $A_g(\lambda)$  modules: the strongly regular case, Duke Math. J. **96** (1999), no. 3, 521–546. MR 1671213 (2000a:22023)
- [Sa57] I. Satake, The Gauss-Bonnet theorem for V-manifolds, J. Math. Soc. Japan 9 (1957), 464–492. MR 0095520 (20 #2022)
- [Sch74] P.A. Schweitzer, Counterexamples to the Seifert conjecture and opening closed leaves of foliations, Ann. of Math. (2) 100 (1974), 386–400. MR 0356086 (50 #8557)
- [Se67] R.T. Seeley, Complex powers of an elliptic operator, Singular Integrals (Proc. Sympos. Pure Math., Chicago, Ill., 1966), Amer. Math. Soc., Providence, R.I., 1967, pp. 288–307. MR 0237943 (38 #6220)

- [Sel56] A. Selberg, Harmonic analysis and discontinuous groups in weakly symmetric Riemannian spaces with applications to Dirichlet series, J. Indian Math. Soc. (N.S.) 20 (1956), 47–87. MR 0088511
- [Sel60] A. Selberg, On discontinuous groups in higher-dimensional symmetric spaces, Contributions to function theory (internat. Colloq. Function Theory, Bombay, 1960), Tata Institute of Fundamental Research, Bombay, 1960, pp. 147–164. MR 0130324
- [Sh18] S. Shen, Analytic torsion, dynamical zeta functions, and the Fried conjecture, Anal. PDE 11 (2018), no. 1, 1–74. MR 3707290
- [Sh20a] S. Shen, Complex valued analytic torsion and dynamical zeta function on locally symmetric spaces, arXiv:2009.03427 (2020).
- [Sh20b] S. Shen, Analytic torsion, dynamical zeta function, and the Fried conjecture for admissible twists, arXiv: 2009.04897 (2020).
- [ShY17] S. Shen and J. Yu, Flat vector bundles and analytic torsion on orbifolds, arXiv: 1704.08369 (2017), to appear in Comm. Anal. Geom.
- [ShY18] S. Shen and J. Yu, Morse–Smale flow, Milnor metric, and dynamical zeta function, arXiv: 1806.00662 (2018), to appear in J. Éc. polytech. Math.
- [Sm67] S. Smale, Differentiable dynamical systems, Bull. Amer. Math. Soc. 73 (1967), 747–817. MR 0228014 (37 #3598)
- [Sp20] P. Spilioti, Twisted Ruelle zeta function and complex-valued analytic torsion, arXiv:2004.13474 (2020).
- [V84] D.A. Vogan, Jr., Unitarizability of certain series of representations, Ann. of Math. (2) 120 (1984), no. 1, 141–187. MR 750719 (86h:22028)
- [VZu84] D.A. Vogan, Jr. and G.J. Zuckerman, Unitary representations with nonzero cohomology, Compositio Math. 53 (1984), no. 1, 51–90. MR 762307 (86k:22040)
- [Vo87] A. Voros, Spectral functions, special functions and the Selberg zeta function, Comm. Math. Phys. 110 (1987), no. 3, 439–465. MR 891947
- [W49] A. Weil, Numbers of solutions of equations in finite fields, Bull. Amer. Math. Soc. 55 (1949), 497–508. MR 0029393
- [Wi66] F.W. Wilson, Jr., On the minimal sets of non-singular vector fields, Ann. of Math. (2) 84 (1966), 529–536. MR 0202155 (34 #2028)
- [Wo08] A. Wotzke, Die Ruellesche Zetafunktion und die analytische Torsion hyperbolischer Mannigfaltigkeiten, Ph.D. thesis, Bonn, Bonner Mathematische Schriften (2008), no. Nr. 389.

### Shu Shen

Institut de Mathématiques de Jussieu-Paris Rive Gauche Sorbonne Université, 4 place Jussieu F-75252 Paris Cedex 5, France

e-mail: shu.shen@imj-prg.fr